

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Dessein, Wouter

Working Paper

Network competition in nonlinear pricing

WZB Discussion Paper, No. FS IV 00-22

Provided in Cooperation with:

WZB Berlin Social Science Center

Suggested Citation: Dessein, Wouter (2000): Network competition in nonlinear pricing, WZB Discussion Paper, No. FS IV 00-22, Wissenschaftszentrum Berlin für Sozialforschung (WZB), Berlin

This Version is available at: https://hdl.handle.net/10419/51025

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



discussion papers

FS IV 00 – 22

Network Competition in Nonlinear Pricing

Wouter Dessein Université Libre de Bruxelles

December 2000

ISSN Nr. 0722 - 6748

Forschungsschwerpunkt Marktprozeß und Unternehmensentwicklung

Research Area Market Processes and Corporate Development

Zitierweise/Citation:

Wouter Dessein, Network Competition in Nonlinear Pricing, Discussion Paper FS IV 00-22, Wissenschaftszentrum Berlin, 2000.

Wissenschaftszentrum Berlin für Sozialforschung gGmbH, Reichpietschufer 50, 10785 Berlin, Tel. (030) 2 54 91 – 0 Internet: www.wz-berlin.de

ABSTRACT

Network Competition in Nonlinear Pricing

by Wouter Dessein

Previous research has argued that, in the mature phase of competition, telecommunications networks may use access charges as an instrument of collusion. We show that this result depends totally on the assumption of linear pricing. Though under nonlinear pricing, the access charge alters the way networks use menus of tariffs to discriminate implicitly among heterogeneous customers, profits are then independent of the access charge, or, if participation constraints are binding, are maximized by the welfare maximizing access charge. In the entry phase, networks often differ in cost structure. An access markup then affects the level playing field between networks.

Keywords: Telecommunications, Interconnection, Two-way Access, Competition Policy, Nonlinear Pricing

ZUSAMMENFASSUNG

Netzwettbewerb bei nichtlinearer Preispolitik

Die industrieökonomische Forschung hat bisher gezeigt, daß in der Marktphase der Reife, beim Wettbewerb von Telekommunikationsnetzen die Zugangsgebühren als ein Instrument des kollusiven Verhaltens eingesetzt werden können. Es wird gezeigt, daß dieses Ergebnis vollständig von den Annahmen des linearen Preisverhaltens abhängt. Obwohl bei nichtlinearer Preispolitik Zugangsgebühren den Einsatz von Tarifmenüs der Netzwerkbetreiber hinsichtlich einer impliziten Diskriminierung zwischen heterogenen Nachfragern beeinflussen, sind die Gewinne dann aber unabhängig von den Zugangsgebühren oder, wenn Teilnahmebeschränkungen bindend sind, dann werden sie maximiert durch die wohlfahrtmaximierenden Zugangsgebühren. Markteintrittsphase sind Netze oft durch Unterschiede in der Kostenstruktur gekennzeichnet. Ein Zugangsgebührenzuschlag beeinflußt dann das Niveau, auf dem der Wettbewerb der Netzanbieter stattfindet.

1 Introduction

Most industrialized countries are engaged in a project to create competition in one of the largest noncompetitive industries of modern economies: local telecommunications. Due to technological advances, the local network has ceased to be a natural monopoly and competing operators are starting to develop their own local and interurban networks, often using cable or wireless technologies. This is likely to affect considerably the way the industry operates. As customers of different operators want to get in touch with each other, networks must be interconnected: there is need for mutual provision of access. We are thus faced with a two-way access problem in which each competitor owns a bottleneck, its subscriber base: a case which differs considerably in its nature from the more familiar one-way access situation where an (integrated) monopolist controls the local network and is required to interconnect with entrants competing on complementary segments such as long distance, value added services,... Whereas in the latter case, the economic literature and practice has made clear that regulation is necessary, this is less obvious for the two-way bottleneck situation. No consensus has yet been reached on this point and particular rules and principles for the setting of two-way interconnection fees are still to be developed in many countries and are evolving in others. Should access charges be freely negotiated between operators as in New Zealand, or should there be a (strong) regulatory involvement as is currently advocated by the FCC in the US and Oftel in the UK? Should private agreements between local operators be trusted to bring about effective competition?

Up to now, the economic literature on two-way access charges has advocated the regulatory approach or at least taken a very ambivalent position. We limit ourselves to a discussion of the seminal paper of Laffont Rey Tirole (1998a) (LRT hereafter).² They present a duopoly model in which customers must decide which network to join and given this choice, how much to call. Per call that terminates off-net, an operator pays a - regulated or negotiated - access charge to its rival. LRT argue that, with linear prices, this access charge may be used as an instrument of collusion due to a raise-each-other's-cost effect: for given market shares, the average perceived marginal cost of a call increases with the access charge so that a higher access charge induces the networks to set a higher retail price. Under appropriate conditions, there exists an access charge which implements the monopoly price as a competitive equilibrium. On the other hand, an access charge below marginal cost may off-set the market power of the networks

¹See, e.g., Laffont-Tirole (1996).

²Independently, Armstrong (1998) obtained similar results for the case of linear pricing. Other papers have built on the basic framework of LRT: Laffont-Rey-Tirole (1998b) analyzes how competition is affected when networks can charge a different price for calls terminating off-net, Doganoglu and Tauman (1996) allow the access charge to depend linearly on the retail prices, whereas Carter and Wright (1999) study the effects of brand loyalty.

and implement the Ramsey price which maximizes social welfare subject to the industry breaking even. Nevertheless, LRT indicate some limits on the collusive power of the access charge. First, it is shown that no equilibrium exists when access charges and/or the substitutability between networks are too large. Second, one of the most striking results of LRT is the dichotomy between competition in linear and nonlinear prices. The collusionary power of the access charge disappears completely when networks compete in two-part tariffs. An access charge above marginal cost still boosts final usage prices, but the positive effect from this on retail profits is totally neutralized by a lower fixed fee. Intuitively, nonlinear pricing erodes the fat profits generated by a high access charge, as networks then have an instrument to build market share without inflating their outflow. It seems, however, quite extreme that in the end, there is no net effect on profits. As nonlinear prices with known demand and homogeneous customers are very particular, and the literature on nonlinear pricing has shown that once customer heterogeneity is introduced, results in general resemble those obtained with linear prices (e.g., usage fees are set different from marginal cost, some surplus is left to customers...), LRT conjecture that the collusion result is likely to be partially restored if one would generalize the model "so as to allow consumers to differ not only in their relative preference for networks but also in their taste for variable consumption." Given that nonlinear pricing is prevailing in the industry, the main question which the literature tries to answer, whether effective competition between telecommunications networks is possible in a deregulated environment, remains thus open.

The aim of this paper is therefore to investigate more thoroughly competition in nonlinear pricing. We stay as close as possible to the basic framework of LRT, but introduce heterogeneity in volume demand in order to have 'more realistic' nonlinear tariffs, that is to allow for second degree price discrimination in which operators use a menu of tariffs to discriminate implicitly between customers of different types. Such heterogeneity in demand is a very important feature of the telecommunications industry. Operators, for example, have in general at least three customer divisions, respectively focusing on the residential, the business and the corporate sector, and also inside these customer categories, especially the residential segment, customers may differ tremendously in their demand, ranging from those who use the telephone only occasionally and for strictly practical matters to the high volume users for whom the telephone is the principal way of communication with the rest of the world. We model this heterogeneity in a straightforward way by assuming that customers are either heavy or light users. As in LRT, however, we still assume a balanced calling pattern: for equal usage prices, each customer receives as much calls as he originates. This assumption is particularly strong given that customers differ in their volume demand: heavy users, e.g., often tend to call more than they are called. Our companion paper,

³Laffont, Rey and Tirole, 1998a; b p.22; p.53

Dessein 1999, shows how competition is affected in the presence of unbalanced calling pattern. Besides heterogeneity between customers, the paper also analyzes heterogeneity among firms and its impact on competition. Such a heterogeneity is especially relevant during the transition towards competition, when new operators often enter the market with a partial coverage and/or a technology different from the incumbent.

As a benchmark, we first pinpoint why, in a symmetric equilibrium and if networks can discriminate explicitly, the access charge has no impact on profits. To understand this, it is useful to recall the classic Hotelling model with unit demands. The latter illustrates the typical oligopoly trade-off between maximizing market share, which calls for a low tariff, and maximizing profits per customer, which calls for a high tariff. An important result is that profits only depend on the extent of the differentiation between the firms and thus not on the cost or quality of the good offered. Network competition in nonlinear tariffs can be seen as a variant of the Hotelling model with unit demands, where the good sold is the membership to a network and its quality is determined by the quantity of telecommunications services that will be consumed. In a symmetric equilibrium, the tariff which networks charge has no effect on access payments between networks; its role is thus identical to the one in the standard Hotelling model: trade-off between maximizing profits per customer and maximizing market share. The only impact of an increase in the access charge is thus to lower the utility offered to customers (the access charge raises the perceived marginal cost of a call which makes it optimal to offer a lower quantity). As the standard oligopoly trade-off is not affected by this utility, it follows that profits are independent of the access charge.⁴

Second, we analyze whether this result still holds if networks cannot discriminate explicitly, and tariffs have a new role in helping discriminate implicitly between heavy and light users. For an access charge equal to the marginal cost of terminating a call, differences between customers may be overcome with a menu of tariffs and it is as if networks could discriminate explicitly. An increase in the access charge, however, lowers the quantities offered in equilibrium and makes the largest offered quantity, destined to heavy users, more and more attractive to light users. For a high enough access charge, light users then prefer the tariff and quantity offered to the heavy users under explicit discrimination, such that in order to discriminate implicitly, networks have to increase the tariff for heavy users and/or lower the tariff for light users. Since the profit function is concave and symmetric on both segments, networks deviate by the same amount on each segment from their best response (corrected for the size of the segment). As a result, a tariff cut is given to the light users (resulting in lower profits on this segment), which is exactly compensated by higher tariffs and profits on the heavy

⁴A similar intuition for the neutrality on profits of the access charge is given in Laffont et al. (1997).

users: there is no net effect on total profits. Although this 'neutrality' of the access charge with respect to profits probably depends on the specific model, it illustrates very well the fact that the access charge has no clear-cut impact on profits. Intuitively, the principal effect of an access mark-up is that it makes it optimal for both networks to offer lower quantities. Once networks compete in nonlinear pricing, however, there is by definition no link anymore between the quantity offered to customers and the way the resulting surplus is shared with them (which determines final profits). While the need to meet certain incentive constraints makes that a new role for offered quantities and tariffs is to help discriminate between customers of different types, concerns for incentive compatibility never make networks use the quantity instrument to share surplus with customers, as in the case of linear pricing. In this sense, agreeing to offer low quantities to customers (by way of agreeing on a high access charge) cannot soften competition for market share.

That the impact of the access charge on profits is ambiguous in more complex settings, is supported by an extension of the model which supposes that customers of different types, perceive the substitutability of the networks in a different way. These different perceived substitutabilities may correspond to different switching costs, different brand loyalty, a differentiated access to publicity and information... Whether an access markup increases or decreases profits then depends on parameter values: when networks are seen as better substitutes by the heavy users than by the light users, networks always obtain higher profits by agreeing on an access charge below marginal cost. In the opposite case, an access charge above marginal cost may boost profits.

Third, we look what happens if some customers may not subscribe a network in equilibrium. We show that in a simple model with homogeneous customers, an access charge different from the marginal cost of terminating a call then leads to more exclusion and lower profits, so that networks strictly prefer an access charge equal to marginal cost.

While the latter results suggest that the industry is competitive in the mature phase of the industry, we argue that this is not always the case during the transition towards competition, when heterogeneity among firms may yield opportunities for anti-competitive access pricing by a dominant network. Entrants may be more (less) efficient than the incumbent and often have access to new technologies (e.g. cable) which differ in their cost structure from the one used by the incumbent. Entrants also tend to have only a partial coverage, so that the incumbent does not face competition for all his customers. Both differences in cost structure and coverage yield asymmetric equilibria. A major implication of the latter are that access revenues are strictly positive or negative: one network owes net access payments to the other. An increase in the access charge then affects the level playing field between networks: it increases the access deficit (revenue) of the network which sets a lower (higher) usage fee. Concretely, we show that an access markup decreases profits of the network with the smallest

variable cost, whereas it increases those of the other network. Our analytical results are calibrated by some simulations which show that even for small differences in cost structure, the impact on profits is substantial. A direct consequence of our results is that the negotiated access charge is likely to reflect the bargaining power of the networks. It is, for example, in the interest of an incumbent with a high variable cost to insist on a high access charge. While ex post, this leads to above marginal cost pricing, ex ante, the prospect of high access charges may deter entry by a low marginal cost entrant. This suggests that even in the absence of collusion, a regulator may play an important role.

This paper is organized as follows. Section 2 and 3 describe our model of heavy and light users under nonlinear pricing. Section 4 discusses the benchmark case of explicit price discrimination in a symmetric equilibrium in which all customers subscribe to a network. Section 5 analyzes implicit price discrimination. Section 6 looks at the impact of limited participation. Section 7, finally, investigates entry and asymmetric equilibria. Section 8 concludes.

2 A model of heavy and light users

We consider the competition between two horizontally differentiated networks. The main elements are as follows:

Cost structure: The two networks have the same cost structure. Serving a customer involves a fixed cost f. Per call, a network also incurs a marginal cost c_o at the originating and terminating ends of the call and a marginal cost c_1 in between. The total marginal cost is thus

$$c = 2c_0 + c_1$$

Demand structure: The networks are differentiated à la Hotelling. Consumers are uniformly located on the segment [0,1] and networks are located at the two extremities, namely at $x_1 = 0$ and $x_2 = 1$. Given income y and telephone consumption q, a type k-consumer located at x joining network i has utility:

$$y + u_k(q) + v_o - \tau |x - x_i|$$

where v_o represents a fixed surplus from being connected⁵, $\tau |x - x_i|$ denotes the cost of not being connected to its "most preferred" network, and the variable gross surplus, $u_k(q)$, is given by:

$$u_k(q) = \frac{k^{\frac{1}{\eta}}q^{1-\frac{1}{\eta}}}{1-\frac{1}{\eta}}$$

 $^{^5}$ We will assume throughout most of the paper that v_o is "large enough", so that all consumers are connected in equilibrium

Faced with a usage fee p, a customer consumes thus a quantity q_k given by

$$u'_k(q_k) = p \Leftrightarrow q_k = kp^{-\eta} \equiv kq(p)$$

Throughout the paper, we will say that the usage fee is p if customers consume a quantity $q_k = kq(p)$. Given that networks compete in nonlinear tariffs, this does not imply that customers effectively face a tariff of the form t(q) = F + pq. We consider two different customer types or customer segments:

- light users, fraction μ of the market, characterized by $k = k_L$.
- heavy users, fraction 1μ of the market, characterized by $k = k_H > k_L$.

The distribution of customers on the segment [0, 1] is assumed to be independent of their type k.

Calling patterns: We suppose that a fraction ℓ of the calls terminates on the light user segment. For the purpose of this paper, we assume that ℓ is such that the calling pattern is balanced:

Definition 1 A calling pattern is balanced whenever for equal usage fees, each customer calls as much as he is being called.

With homogeneous customers, this is realized very naturally by assuming that customers receive the same amount of calls ($\ell = \mu$). With heterogeneous customers, a different assumption is needed:

Lemma 1 A calling pattern is balanced if and only if

$$\ell = \frac{\mu k_L}{\mu k_L + (1 - \mu)k_H}.$$

Proof. For equal usage fees, a light user receives the same amount of calls as he originates if and only if

$$\ell \left[\mu k_L q(p) + (1 - \mu) k_H q(p) \right] = \mu k_L q(p) \Leftrightarrow \ell = \frac{\mu k_L}{\mu k_L + (1 - \mu) k_H}$$

By construction, the same holds for heavy users.

Given that customers differ in their volume demand, the assumption of a balanced calling pattern is quite strong. Heavy users, for example, often originate more calls than they receive. Our companion paper, Dessein 1999, analyzes the case in which customers of one type tend to call more than they are being called.

3 Competition in nonlinear pricing

Competition in optimal nonlinear pricing is investigated, both under the assumption that networks can discriminate explicitly between heavy and light users (third-degree price discrimination) and in the more realistic case in which only implicit discrimination (second-degree price discrimination) is allowed. From the revelation principle, networks cannot do better than offering a quantity q_L for a tariff t_L and a quantity q_H for a tariff t_H where, under implicit discrimination, $\{q_L, t_L, q_H, t_H\}$ must be such that light users opt for (q_L, t_L) and heavy users choose (q_H, t_H) . The variable net surplus of respectively a light and a heavy users is thus

$$w_L \equiv u_{k_L}(q_L) - t_L$$
 and $w_H \equiv u_{k_H}(q_H) - t_H$

For given net surpluses (w_L, w_H) and (w'_L, w'_H) offered by network 1 and 2, the market shares α_L and α_H of network 1 in respectively the light and the heavy users' segment are determined as in Hotelling's model. A consumer of type s (s = L, H) located at $x = \alpha_s$ is indifferent between the two networks if and only if

$$w_s - \tau \alpha_s = w_s' - \tau (1 - \alpha_s),$$

or

$$lpha_s = lpha(w_s, w_s') \equiv rac{1}{2} + \sigma \left[w_s - w_s'\right]$$

$$\sigma \equiv rac{1}{2\sigma}$$

where

is an index of substitutability between the two networks. Given our assumptions about the calling pattern, the *share in incoming calls* of network 1 is

$$\hat{\alpha} = \alpha_L \ell + \alpha_H (1 - \ell)$$

Let a denote the unit access charge to be paid for interconnection by a network to its competitor. Network 1's profits are

$$\pi = \mu \alpha_L \left[t_L - (c + (1 - \hat{\alpha})(a - c_o)) q_L - f \right] +$$

$$(1 - \mu) \alpha_H \left[t_H - (c + (1 - \hat{\alpha})(a - c_o)) q_H - f \right] +$$

$$\left[\mu (1 - \alpha_L) q'_L + (1 - \mu)(1 - \alpha_H) q'_H \right] \hat{\alpha}_1 (a - c_o)$$
(1)

These profits can be decomposed into a retail profit

$$\mu\alpha_L \left[t_L - cq_L - f\right] + (1 - \mu)\alpha_H \left[t_H - cq_H - f\right]$$

which would be made if all calls terminated on net, plus an access revenue

$$A_{1} = \left[\mu(1 - \alpha_{L})q'_{L} + (1 - \mu)(1 - \alpha_{H})q'_{H} \right] \hat{\alpha}(a - c_{o}) - \left[\mu\alpha_{L}q_{L} + (1 - \mu)\alpha_{H}q_{H} \right] (1 - \hat{\alpha})(a - c_{o}).$$

From our balanced calling pattern assumption, regardless of α_L and α_H , $A_1 = 0$ if all customers face the same usage fee.

4 Explicit price discrimination

As a benchmark, we analyze competition when networks can explicitly discriminate customers, so that customers are forced to use the tariff which is destined for their type. As market shares only depend on the variable net surplus, it is convenient to view networks as picking quantities (q_H, q_L) and net surpluses (w_H, w_L) rather than quantities and tariffs (t_H, t_L) . Network 1's profits are then

$$\pi = \mu \alpha_L \left[\frac{\eta}{\eta - 1} k_L^{1/\eta} q_L^{1 - 1/\eta} - w_L - c q_L - f \right]$$

$$+ (1 - \mu) \alpha_H \left[\frac{\eta}{\eta - 1} k_H^{1/\eta} q_H^{1 - 1/\eta} - w_H - c q_H - f \right] + A_1$$
(2)

From the first order conditions with respect to q_L and q_H , the equilibrium usage fees equal the average perceived marginal cost of a call, $c + (1 - \hat{\alpha})(a - c_o)$. In a symmetric equilibrium,⁶ we have

$$\hat{q}_L = k_L q(c + \frac{a - c_o}{2})$$
 and $\hat{q}_H = k_H q(c + \frac{a - c_o}{2})$

A higher access charge thus makes it optimal for networks to offer lower quantities and, for $a \geq c_o$, lowers the total surplus created in the industry. Intuitively, a firm offers its customers the quantity which maximizes the sum of its and their surplus, and thus includes the access losses generated by these customers. Whereas the role of the offered quantities is to maximize the joint surplus, the role of the transfers is to share this surplus with customers. Given that all customers face the same usage fee in equilibrium, access revenues are always zero and the joint surplus per customer is independent of market shares and tariffs; hence, when setting t, networks face the classic Hotelling trade-off between: (a) maximizing market share, which calls for a low tariff; (b) keeping as much as possible from the per customer surplus, which calls for a high tariff. As in the standard Hotelling model, profits on a customer segment of size μ can be written as

$$\pi \equiv \mu \alpha(w, w') \bar{\pi} = \mu \left[\frac{1}{2} + \sigma(w - w') \right] \bar{\pi} = \mu \left[\frac{1}{2} + \sigma \left(S - \bar{\pi} - \left(S - \bar{\pi}' \right) \right) \right] \bar{\pi}$$

where S = S' is the total surplus per customer, $\bar{\pi}$ and $\bar{\pi}'$ are the profits per customer and w and w' are the offered net surpluses of network 1 and 2. It follows that profits are independent of the total surplus per customer. Since this total surplus is the only thing which is directly effected by the access charge and the customer type, profits are thus also independent of a and k. Maximizing (1) with respect to w_L and w_H , we have

$$\hat{t}_L = 1/2\sigma + f + c\hat{q}_L
\hat{t}_H = 1/2\sigma + f + c\hat{q}_H,$$

from which profits per firm are $1/4\sigma$, as in the Hotelling model.

⁶Following LRT, it can be shown that for a close to c_o , there exists an equilibrium which is unique and symmetric.

Proposition 1 If networks discriminate explicitly, profits per customer are independent of the access charge and the customer type. The usage fees are equal to $p = c + \frac{a-c_o}{2}$ for both heavy and light users.

Note that though usage fees equal the perceived marginal cost, the difference in final tariffs for \hat{q}_H and \hat{q}_L reflect the true incremental cost $c(\hat{q}_H - \hat{q}_L)$.

Remark: A variant of our model which also comprises the Hotelling model with unit demands as a special case, could let the differentiation of the networks depend on the quantity consumed in equilibrium, so that, for example, a type k-consumer located at x joining network i has utility

$$y + u_k(q) + v_o - \tau |x - x_i| q$$

Given that an increase in the access charge makes it optimal for networks to offer lower quantities, it then also increases the substitutability of the networks and thus competition for market share. Profits are therefore likely to decrease with the access charge if the differentiation does not take the form of a fixed benefit which customers receive when joining a particular network.

5 Implicit price discrimination

If customers are homogeneous or if explicit price discrimination is possible, price schedules must achieve two goals: creating surplus and sharing it with customers. Armstrong (1998) and Laffont, Rev and Tirole (1998a) have shown that when networks have only *one* instrument, a price per unit, a higher access charge then softens competition for market share and can thus be used to boost profits. Under nonlinear pricing, however, networks have two instruments (quantities and tariffs) which allow them to separate the two roles of pricing. As the access charge only has an impact on the optimal quantity to be offered to customers, and not on the competition for market share (how much of the surplus to share with customers), profits are unaffected. If customers are heterogeneous and networks are not allowed to discriminate explicitly, a new, third, role of pricing is to discriminate implicitly between customers of different types. The literature on nonlinear pricing has then shown then that results in general resemble those obtained with linear prices (for example, usage fees are set above marginal cost, some surplus is left to customers). Both Armstrong (1998) and Laffont, Rey Tirole (1998a,b) conjecture therefore that the collusion result, obtained under linear pricing, is likely to be partially restored once there are differences in volume demand for consumption which cannot completely be overcome with a menu of tariffs. The central question of this section is thus whether the difference between the number of goals and instruments under implicit price discrimination indeed allows

⁷See, for example, Varian (1989) and Wilson (1992).

networks to use a positive access markup to increase their profits, as is the case under linear pricing.

5.1 Standard model

If networks are not allowed to price discriminate explicitly according to whether a customer is a heavy or a light user (that is third-degree price discrimination is ruled out), the proposed menu of tariffs $\{q_L, t_L, q_H, t_H\}$ must be such that heavy users opt for (q_H, t_H) and light users choose (q_L, t_L) . The incentive constraints (IC) are

$$w_H = u_{k_H}(q_H) - t_H \ge u_{k_H}(q_L) - t_L$$
 (IC_H)

$$w_L = u_{k_L}(q_L) - t_L \ge u_{k_L}(q_H) - t_H$$
 (IC_L)

If $a=c_o$, the quantities offered under explicit price discrimination are those generated by usage fees set at marginal cost, while tariffs are so that profits per customer equal $1/2\sigma$. But this tariff structure is incentive compatible; it can for example be implemented by a unique two-part tariff, t(q)=pq+F, in which p=c and $F=1/2\sigma+f$.⁸ As IC_H and IC_L are both satisfied with strict inequality whenever $k_L \neq k_H$ and thus $\hat{q}_L \neq \hat{q}_H$, the explicit discrimination equilibrium is still incentive compatible for a close to c_o .⁹

For a large or k_L close to k_H , however, $\{\hat{q}_L, \hat{t}_L, \hat{q}_H, \hat{t}_H\}$ is no longer incentive compatible. Indeed, the explicit price discrimination equilibrium is given by $\hat{q}_s = k_s \hat{q}$ and $\hat{t}_s = \hat{t}(\hat{q}_s)$, (s = L, H), where

$$\hat{q} = q(c + \frac{a - c_o}{2}) \tag{3}$$

$$\hat{t}(q) = 1/2\sigma + f + cq. \tag{4}$$

Given (4), the net utility of a customer of type k is concave in q and reaches a maximum for q = kq(c). From (3), an increase in the access charge lowers both \hat{q}_L and \hat{q}_H and, for large enough an access charge, $\hat{q}_L < \hat{q}_H \le k_L q(c)$. Light users then strictly prefer (\hat{q}_H, \hat{t}_H) to (\hat{q}_L, \hat{t}_L) and the IC of the light users is violated.

⁸This is in line with the results of Armstrong (1998) and Rochet and Stole (1998) which show that simple two-part tariffs often arise in competitive environments where consumers have private information about their tastes. Rochet and Stole (1998) show, however, that this result is highly sensitive to the assumption that the customer's type is uncorrelated with the consumers location on the Hotelling line and that all consumer types are willing to participate with the candidate tariffs.

⁹Notice, however, that the equilibrium contract $\{\hat{q}_L, \hat{t}_L, \hat{q}_H, \hat{t}_H\}$ then cannot be implemented through two-part tariff(s). In order for customers to chose the optimal quantities \hat{q}_L and \hat{q}_H , the usage fee for both types must be set at $p_L = p_H = c + \frac{a - c_o}{2}$; to implement \hat{t}_L and \hat{t}_H , however, a different fixed fee is needed for heavy and light users, which, of course, is not incentive compatible.

Similarly, for $a > c_o$, $\hat{q}_L < k_L q(c)$ so that for k_L close enough to k_H ,

$$\hat{q}_L < \hat{q}_H = \frac{k_H}{k_L} \hat{q}_L \le k_L q(c),$$

and again the IC of the light users is violated. The polar case is obtained for $a < c_o$: for a small enough or k_L close enough to k_H , $k_H q(c) \le \hat{q}_L < \hat{q}_H$ and the IC of the heavy users is violated. The reason behind the impact of the access charge on incentive compatibility is thus simple: tariffs for - and thus preferences over - \hat{q}_L and \hat{q}_H reflect their true marginal cost c, while \hat{q}_L and \hat{q}_H themselves are determined by the perceived marginal cost $c + (a - c_o)/2$.

Intuitively, the IC which is violated under explicit discrimination, will be binding in the equilibrium under implicit discrimination. The next lemma guarantees this for $\delta = k_H - k_L$ small.

Lemma 2 If the explicit price discrimination equilibrium, $\{\hat{q}_L, \hat{t}_L, \hat{q}_H, \hat{t}_H\}$ violates the incentive constraint of the light (heavy) users, then for $\delta = k_H - k_L$ small, any symmetric equilibrium $\{t_L^*, q_L^*, t_H^*, q_H^*\}$ under implicit price discrimination is such that the incentive constraint of the light (heavy) users is binding.

Proof. See appendix $9.1.1 \blacksquare$

When the IC of the light users is binding, the symmetric equilibrium is characterized by $\{q_L^*, t_L^*, q_H^*, t_H^*\}$, where

$$q_L^* = \hat{q}_L; \quad t_L^* < \hat{t}_L$$

 $q_H^* > \hat{q}_H; \quad t_H^* > \hat{t}_H + c(q_H^* - \hat{q}_H).$

By increasing q_H and $(t_H - t_L)$, networks make $\{\hat{q}_H, \hat{t}_H\}$ less attractive to light users. Note that by providing a larger quantity to heavy users than is optimal given the perceived marginal cost, networks eliminate partly the distortion induced by this inflated perceived marginal cost. This is in contrast with standard results of nonlinear pricing where implicit price discrimination lowers the offered quantity and reduces efficiency. More important, however, compared to the explicit price discrimination equilibrium, a higher surplus is left to the light users. While this lowers profits made on these customers, average profits are nevertheless unaffected, as the loss is exactly compensated by higher profits on the heavy users.

Proposition 2 • In a symmetric equilibrium, profits are equal to $1/4\sigma$, irrespective of the access charge.

• Fix the average customer type k and let the difference $\delta = k_H - k_L$ vary. For any δ_o , there exists an access charge $a_o > c_o$ such that a symmetric equilibrium always exists for $a \leq a_o$ and $\delta \leq \delta_o$. Moreover,

- For any given $\delta \leq \delta_o$, for a close to c_o , incentive constraints are non-binding and the equilibrium is the same as if networks could explicitly discriminate between heavy and light users.
- For any given a ∈]c_o, a'], for δ close to 0, the incentive constraint of the light users is binding. Compared to the equilibrium with explicit price discrimination: (1) the quantity (and tariff) offered to heavy users is higher, (2) a lower tariff for an unchanged quantity is charged to light users.

Proof. See appendix $9.1.1 \blacksquare$

For δ small enough, an access charge above marginal cost induces the IC of light users to be binding and could thus boost profits if incentive compatibility constraints lead networks to increase their tariffs. We give the intuition for why this does not arise here. Starting from the explicit price discrimination equilibrium, networks must make (t_H, q_H) less attractive and/or (t_L, q_L) more attractive to the light users to meet their IC. Besides changes in the offered quantities, networks are obliged to decrease $t_L - t_H$. Neglecting incentive constraints, given the equilibrium quantities q_H^* and q_L^* and given the tariffs t_H' and t_L' charged by network 2, network 1's profits on a segment s (s = L, H) are concave in t_s and reach a maximum for

$$\hat{r}(t'_s) = \frac{\hat{t}(q^*_s) + t'_s}{2}$$
 where $\hat{t}(q^*_s) = 1/2\sigma + f + cq^*_s$

In order to meet the incentive constraints, it is thus never optimal to decrease only t_L or raise only t_H relative to $\hat{r}(t_L')$ and $\hat{r}(t_H')$: network 1 will spread its losses and deviate least from $\hat{r}(t_s')$ on the segment s where the profit function is most concave. In our model, the concavity of the profit function on a segment depends only on the substitutability, (the higher σ , the larger is the impact of changes in t on market shares and thus on profits) and on the size of the segment (the more customers are in a segment the more expensive it is to deviate from the 'optimal' per customer tariff). Hence, if only changes in $t_L - t_H$ matter, network 1 always deviates from its best response in a way that

$$\mu \left[t_H - \hat{r}(t_H') \right] = (1 - \mu) \left[\hat{r}(t_L') - t_L \right]$$

$$\Leftrightarrow \mu \left[t_H - \frac{\hat{t}(q_H^*) + t_H'}{2} \right] = (1 - \mu) \left[\frac{\hat{t}(q_L^*) + t_L'}{2} - t_L \right]$$

$$\frac{\partial^2 \pi}{\partial t_I^2} = -2\mu\sigma \qquad \frac{\partial^2 \pi}{\partial t_H^2} = -2(1-\mu)\sigma$$

That the second derivative is independent of t_s is specific to our model and stems from the fact that profits are quadratic/market shares are linear in t.

 $^{^{10}}$ Indeed, for given equilibrium quantities, the second derivative of profits with respect to t_L and t_H are

In a symmetric equilibrium, $t_L = t'_L$ and $t_H = t'_H$, this yields

$$\mu \left[t_H - \hat{t}(q_H^*) \right] = (1 - \mu) \left[\hat{t}(q_L^*) - t_L \right] \tag{5}$$

As for $\{q_H^*, \hat{t}(q_H^*), q_L^*, \hat{t}(q_L^*)\}$, profits equal $1/2\sigma$ per customer, average per customer profits given $\{q_H^*, t_H, q_L^*, t_L\}$ are also $1/2\sigma$. Though the need to discriminate implicitly affect the profits made on each segment, total profits are unaffected.

While this 'neutrality' of the access charge is an extreme result which probably depends on the quadratic profit function, we think its intuition is very robust: the access charge has no clear-cut impact on profits. As a principal effect, the access charge induces networks to offer lower quantities to their customers. Networks who agree on a high access charge thus agree on offering customers low quantities. When the quantity offered to customers is linked to the way the resulting surplus is shared with customers (as with linear pricing), an increase in the access charge may reduce competition for market share and result in higher profits. networks compete in nonlinear pricing, however, there is no such clear link. The need to meet certain incentive constraints means that the role of the offered quantity is not just maximizing joint surplus, and the role of tariffs not just to share this surplus optimally with customers, but also to help discriminate between customers of different types. Concerns for incentive compatibility, however, never lead networks to use the quantity instrument to share surplus with customers. In this sense, agreeing on offering low quantities to customers (by way of agreeing on a high access charge) cannot soften competition for market share.

5.2 Two-part tariffs

Though sophisticated nonlinear tariffs are more and more common in telecommunications, especially in the mobile sector, two-part tariffs are still very popular. Suppose therefore that for an exogeneous reason, networks are limited to the use of (a menu of) two-part tariffs. As there are only two types, it is optimal to let customers choose between

$$t_L(q) = F_L + p_L q$$
 and $t_H(q) = F_H + p_H q$

where $\{F_L, p_L, F_H, p_H\}$ must be such that heavy users opt for (F_H, p_H) and light users choose (F_L, F_L) . Faced with a tariff t(q) = F + pq, the net surplus of a customer of type s (s = L, H) is

$$w_s(p,F) \equiv k_s v(p) - F$$

with

$$k_s v(p) \equiv \max_q \left\{ u_k(q) - pq \right\} = k \left(\frac{p^{-(\eta - 1)}}{\eta - 1} \right)$$

Incentive constraints are then

$$w_H(p_H, F_H) \ge w_H(p_L, F_L)$$
 and $w_L(p_L, F_L) \ge w_L(p_H, F_H)$

Proposition 3 • In any symmetric equilibrium in two-part tariffs, profits are independent of the access charge and are equal to $1/4\sigma$

• For a close to c_o , a symmetric equilibrium exists and is such that for $a > c_o$, the incentive constraint of the light users is binding and

$$- p_L = c + \frac{a - c_o}{2}, \quad F_L < 1/2\sigma + f - \frac{a - c_o}{2}k_L q(p_L) - p_H < c + \frac{a - c_o}{2}, \quad F_H > max\{F_L, 1/2\sigma + f - (p_H - c)k_H q(p_H)\}.$$

Proof. See appendix $9.1.2 \blacksquare$

Proposition 3 shows that the profit neutrality of the access charge does not depend on our assumption of *optimal* nonlinear pricing. Again, networks have two instruments, a usage fee which directly controls the offered quantity and a fixed fee to share surplus with customers. Though the usage price also partially determines how the surplus is shared, its impact can be corrected in a profit-neutral way by changes in the fixed fee. As in the case of optimal nonlinear pricing, agreeing to offer a low quantity to customers has thus no impact on the way the resulting surplus is shared.

5.3 Limits to the neutrality of the access charge on profits.

This section illustrates our claim that, while the access charge may have an impact on profits, this impact is as likely to be positive as negative. Suppose customers of different types perceive the substitutability of the networks in a different way. Such different perceived substitutabilities can correspond to different brand loyalties, different search costs, a differentiated access to product information or publicity, different switching costs, etc. We denote these perceived substitutabilities for light and heavy users respectively by σ_L and σ_H . If networks could discriminate explicitly, they would offer quantities that reflect the perceived marginal cost

$$\hat{q}_L = k_L q(c + \frac{a - c_o}{2}), \quad \hat{q}_H = k_H q(c + \frac{a - c_o}{2})$$
 (6)

and charge a tariff $t(q_s)$ such that profits per customer are $1/2\sigma_s$, (s=L,H):

$$\hat{t}(\hat{q}_L) = 1/2\sigma_L + f + c\hat{q}_L \tag{7}$$

$$\hat{t}(\hat{q}_H) = 1/2\sigma_H + f + c\hat{q}_H \tag{8}$$

Total profits under explicit price discrimination, denoted by π_D^* , are thus independent of the access charge and are equal to

$$\pi_D^* = \frac{\mu}{4\sigma_L} + \frac{1 - \mu}{4\sigma_H}. (9)$$

If explicit price discrimination is not possible, networks still can discriminate implicitly through a menu of tariffs. A first consequence of $\sigma_L \neq \sigma_H$, is that incentive compatibility conditions may affect profits. Denoting profits under implicit price discrimination by π^* , one has

$$\pi^* < \pi_D^* \text{ when } \sigma_H > \sigma_L \text{ and } \pi^* > \pi_D^* \text{ when } \sigma_H < \sigma_L$$
 (10)

if the IC of the light users is binding, and

$$\pi^* > \pi_D^* \text{ when } \sigma_H > \sigma_L \text{ and } \pi^* < \pi_D^* \text{ when } \sigma_H < \sigma_L$$
 (11)

if the IC of the heavy users is binding. The intuition of this result is similar to that of the profit neutrality of the access charge in the standard model. Recall that if an IC is binding, networks must deviate from their best responses under explicit discrimination in order to meet incentive constraints. As only the difference between t_L and t_H matters, and profits are concave in t_L and t_H , networks deviate most on the segment where profits are least concave. As ceteris paribus, profits are least concave on the segment with the smallest substitutability (market shares react much faster when the substitutability is stronger), networks deviate most on the segment where σ is smallest. Depending on whether the needed deviation on that particular segment is a tariff cut or a tariff raise, equilibrium profits are then either lower or higher than under explicit price discrimination. In case, for example, the IC of the light users is binding, networks must decrease $t_H - t_L$. If $\sigma_H > \sigma_L$, networks find it optimal to decrease more t_L than they increase t_H so that profits are lower than under explicit price discrimination. The opposite holds if $\sigma_H < \sigma_L$.

As the access charge affects incentive compatibility, a second implication of $\sigma_L \neq \sigma_H$, is that the access charge may also affect profits.

- Suppose first that $a=c_o$. If Δk is relatively small compared to $|\sigma_H-\sigma_L|$, then the IC of the customers with the smallest perceived substitutability will be violated in the explicit price discrimination equilibrium: from (6),(7) and (8), the difference in equilibrium quantities is then small relative to the difference in equilibrium tariffs. From lemma 2, if Δk is small, the same IC is binding in the implicit price discrimination equilibrium. From (7) and (8), if $\sigma_H > \sigma_L$, the IC of the light users is thus binding for Δk small, from which $\pi^* < \pi_D^*$. Similarly, if $\sigma_H < \sigma_L$, the IC of the heavy users is binding for Δk small, from which also $\pi^* < \pi_D^*$. Of course, if Δk is relatively large, networks can perfectly discriminate and $\pi^* = \pi_D^*$.
- Consider now $a \neq c_o$. In the explicit price discrimination equilibrium, customers of type k are offered a quantity $\hat{q}_k = kq(c + \frac{a-c_o}{2})$ for a tariff $\hat{t}_k = 1/2\sigma_k + f + c\hat{q}_k$. Defining

$$V_k(q) \equiv u_k(q) - cq$$
, and $F_k \equiv 1/2\sigma_k + f$,

the IC of light and heavy users can be rewritten as

$$V_L(\hat{q}_L) - V_L(\hat{q}_H) > F_L - F_H \qquad (IC_L)$$

$$V_H(\hat{q}_L) - V_H(\hat{q}_H) < F_L - F_H$$
 (13)

Note that $V_k(q)$ is strictly concave and reaches a maximum for q = kq(c). We distinguish the impact of a positive and a negative access markup:

- (1) A negative access markup $(a < c_o)$ has a clear impact on IC_L and IC_H . When $c + \frac{a-c_o}{2}$ goes to zero, $\hat{q}_H \hat{q}_L$ tends to infinity such that $V_s(\hat{q}_L) V_s(\hat{q}_H)$, (s = L, H) increases without a bound when a gets smaller. For a small enough, IC_L is then always satisfied, while IC_H is violated. A negative access markup thus strengthens the IC of the heavy users and weakens the IC of the light users. Intuitively, the larger the quantities offered in equilibrium (due to the low access charge), the more everybody likes the smallest offered quantity: the incremental gross utility of consuming \hat{q}_H instead of \hat{q}_L , given by $u(\hat{q}_H) u(\hat{q}_L)$, decreases more and more compared to its incremental cost, $F_H F_L + c(\hat{q}_H \hat{q}_L)$. It follows that for $\sigma_H > \sigma_L$ and Δk small, a sufficiently negative access markup increases profits: the IC of the heavy users is then binding so that $\pi^* > \pi_D^*$, while for $a = c_o$, the IC of the light users is binding and thus $\pi^* < \pi_D^*$.
- (2) A positive access markup $(a > c_o)$, on the other hand, always decreases $V_s(\hat{q}_L) V_s(\hat{q}_H)$ (s = L, H), though not without limits: $V_s(\hat{q}_L) V_s(\hat{q}_H)$ (s = L, H) reaches a negative lowerbound for some $a_H^* > c_o$ and $a_L^* > c_o$. For $|\sigma_H \sigma_L|$, and thus also $|F_H F_L|$ small enough, there exists then an $a > c_o$, such that the IC of the light users is violated by the explicit price discrimination equilibrium. If follows that for $\sigma_H < \sigma_L$, the IC of the light users is binding so that $\pi^* > \pi_D^*$.

Proposition 4 • Suppose $a = c_o$:

- (1) Given $|\sigma_H \sigma_L|$, for Δk sufficiently small, the incentive constraint of customers with the smallest perceived substitutability is binding. Equilibrium profits π^* are then smaller than π_D^* .
- (2) Given Δk , for $|\sigma_H \sigma_L|$ sufficiently small, no incentive constraints are binding and $\pi^* = \pi_D^*$.
- The access charge may affect profits:
 - (1) For a sufficiently negative access markup ($a < c_o$) and Δk small, the incentive constraint of the heavy users is binding. If $\sigma_H > \sigma_L$, then $\pi^* > \pi_D^*$: a sufficiently negative access markup increases profits. If $\sigma_L < \sigma_H$, then $\pi^* < \pi_D^*$.
 - (2) For a given positive access markup $(a' > c_o)$ and Δk small, if $|\sigma_H \sigma_L|$ is sufficiently small, the incentive constraint of the light users is binding. If $\sigma_H < \sigma_L$, then $\pi^* > \pi_D^* : a' > c_o$ boosts profits relative to $a = c_o$. If $\sigma_H > \sigma_L$, $\pi^* < \pi_D^*$ for $a' > c_o$.

Proof. See appendix $9.1.3 \blacksquare$

6 Limited participation

We have assumed so far that the fixed surplus of consuming telephone services, v_o , is large enough, so that in equilibrium customers always subscribe to one of the networks, that is participation constraints are always satisfied. This section shows how, in a simple model in which all customers are of the same type $(k_1 = k_2 = 1)$, our results are affected if one relaxes this assumption. Intuitively, as in the standard model an access charge above marginal cost reduces the net surplus offered to customers, an access markup is likely to result in more exclusion of customers and thus to lower profits if participation constraints are binding. Denote by w_i the net surplus, including the fixed benefit of subscribing, offered by network i to a customer residing at its location:

$$w_i = v_o + u(q_i) - t_i$$

Given the surplus w_j , two regions of network i's demand may be distinguished: a "monopoly" region, where marginal customers' best alternative is not to subscribe and a "competitive" region, where marginal customers are willing to subscribe to the rival network. In the absence of competition of network j (e.g. $w_j = 0$), the demand for network i is

$$\alpha_i = \frac{w_i}{\tau} = 2\sigma w_i$$

This defines the potential market of network i. In the competitive region, network i's demand is given by

$$\alpha_i = \frac{1}{2} + \sigma \left(w_i - w_j \right).$$

Intuitively, if v_o is large enough (as we assumed before), the equilibrium always lies in the competitive region. Note that demand is more elastic in the monopoly region than in the competitive region, so that if $0 < w_j < \tau$, the demand curve of network i is kinked; starting from $w_i = 0$, by offering a higher surplus, network i first captures customers who otherwise would not subscribe at all, and finally captures customers who would otherwise subscribe to network j.

For simplicity, we assume that calls to different customers are perfect substitutes. This implies that demand for calls is always $q(p) \equiv p^{-\eta}$, even if only a fraction ϕ of customers has subscribed a network.¹¹ Define s^* as the surplus of a subscription by a customer residing at the location of network i or j if the usage fee is equal to $c + \frac{a-c_0}{2}$, then

$$s^* = v_o + u(q(c + \frac{a - c_o}{2})) - cq(c + \frac{a - c_o}{2}) - f.$$

Proposition 5 If a symmetric equilibrium exists, then it is such that

¹¹If calls to different customers were not substitutes, demand would equal $q_{\phi}(p) \equiv \phi p^{-\eta}$.

• if
$$s^* > \frac{3}{4\sigma}$$
, then $\alpha_i = 1/2$ and $\pi^* = \frac{1}{4\sigma}$

• if
$$s^* \in \left[\frac{1}{2\sigma}, \frac{3}{4\sigma}\right]$$
, then $\alpha_i = 1/2$ and $\pi^* = \frac{s^*}{2} - \frac{1}{4\sigma} \le \frac{1}{4\sigma}$

• if
$$s^* < \frac{1}{2\sigma}$$
, then $\alpha_i < 1/2$ and $\pi^* = \frac{\sigma(s^*)^2}{2}$

As s^* is concave in a and reaches a maximum for $a = c_o$, symmetric equilibrium profits decrease with $|a - c_o|$ if $s^* \le \frac{3}{4\sigma}$.

Proof. Competition in quantities and transfers can equivalently be viewed as competition in quantities and net surpluses. As usual, it follows from the FOC with respect to q, that offered quantities reflect the perceived marginal cost. In a symmetric equilibrium, we then have

$$q^* = q(c + \frac{a - c_o}{2})$$

The surplus of a subscription of a customer residing at the location of network i or j equals then s^* . The rest of the proof is standard.

From proposition 5, if participation constraints are binding, profits are decreasing in the net surplus that would have been offered to customers if no participation constraints were binding. As the latter is maximal for $a = c_o$, firms strictly prefer an access charge equal to marginal cost, which is also the access charge which maximizes social welfare. Intuitively, by maximizing the total surplus of a subscription, networks are able to make more profits on more customers.

7 Entry and asymmetric equilibria

The previous analysis suggests that there is no need for regulation in the mature phase of the industry. One should be a more cautious, however, in drawing this conclusion during the transition towards competition. Due to asymmetries in cost structure or coverage, the access charge may then have a different impact on the profits of competing networks. This section investigates whether an access charge above marginal cost may give an unfair competitive advantage to one network. For this purpose, we analyze a range of asymmetric equilibria which incorporate the following extensions of our basic model:

(1) Partial entry: In contrast with the incumbent (network 1), an entrant (network 2) often covers only part of the market. Denoting the coverage of the entrant by $\mu \leq 1$, its market share is given by:

$$lpha_2 = \mu \left[rac{1}{2} + \sigma(w_2 - w_1)
ight]$$

where w_i is the variable net surplus offered by network i to his customers.

(2) Differences in cost structure. Networks may differ in their variable cost c_i and/or fixed cost f_i (i = 1, 2). Such differences in costs are typical during the transition towards competition, when operators often enter the market with a technology different from the one used by the incumbent. To avoid the issue of nonreciprocal access charges, however, we maintain the assumption that the cost of terminating a call, c_o , is identical to both networks.

For simplicity, customers are also assumed to be homogeneous $(k_L = k_H = 1)$.

Recall that in a symmetric equilibrium, networks' profits are equal to those of the symmetric Hotelling model with unit demands. Before we proceed with the analysis, we therefore define equilibrium profits and equilibrium market shares in an asymmetric Hotelling model with unit demands. Denoting gross utility minus cost of good i by s_i and network i's per customer profits by R_i , then, if $\mu = 1$, total profits of firm i are

$$\alpha_i R_i = \left(\frac{1}{2} + \sigma(w_i - w_j)\right) R_i = \left(\frac{1}{2} + \sigma\left[s_i - R_i - s_j + R_j\right]\right) R_i,$$

yielding equilibrium profits

$$\pi_i^H = \frac{(\alpha_i^*)^2}{\sigma} \tag{14}$$

and equilibrium market shares

$$\alpha_i^* \equiv \alpha_i^H(s_i, s_j) = \frac{1}{2} + \frac{\sigma}{3} [s_i - s_j]$$
 (15)

If network 2 covers only $\mu < 1$ of the market, then

$$\alpha_1^H(s_1, s_2) = \frac{1}{2} + \frac{1-\mu}{3} + \frac{\mu\sigma}{3}[s_1 - s_2]$$
 (16)

$$\alpha_2^H(s_1, s_2) = \frac{1}{2} - \frac{1-\mu}{3} + \frac{\mu\sigma}{3} [s_2 - s_1]$$
 (17)

The following proposition characterizes all asymmetric shared market equilibria and the impact of the access charge on them.

Proposition 6 • In any shared market equilibrium, profits of network 1 and 2 are

$$\pi_1^* = \pi_1^H + (\alpha_1^*)^2 (q_2^* - q_1^*) (a - c_o)
\pi_2^* = \pi_2^H - (\alpha_2^*)^2 (q_2^* - q_1^*) (a - c_o)$$
(18)

resulting in total industry profits

$$\pi_1^H + \pi_2^H + (\alpha_1^* - \alpha_2^*) (q_2^* - q_1^*) (a - c_o)$$
(19)

where q_i^* and α_i^* are solutions to

$$q(p_i^*) = q(c_i + \alpha_j^*(a - c_o))$$

 $\alpha_i^* = \alpha_i^H(s_i(q_i^*), s_j(q_i^*))$

with $s_i(q_i^*) \equiv u(q_i^*) - c_i q_i^* - f_i$

- Suppose $c_i < c_j$ and define $\tilde{\alpha}_i$, $\tilde{\alpha}_j$ as the equilibrium market shares for $a = c_o$:
 - If $\tilde{\alpha}_i < \tilde{\alpha}_j$, then a small access markup $(a > c_o)$, decreases α_i^* and π_i^* , but increases π_i^* and total industry profits.
 - If $\tilde{\alpha}_i > \tilde{\alpha}_j$, but

$$\left(\frac{\tilde{\alpha}_i}{\tilde{\alpha}_j}\right)^2 < \left(\frac{c_j}{c_i}\right)^{\eta+1},$$
(20)

then a small access markups markup decreases α_i^* , π_i^* and total industry profits, but increases π_i^* .

- If $\tilde{\alpha}_i > \tilde{\alpha}_j$, and

$$\left(\frac{\tilde{\alpha}_i}{\tilde{\alpha}_j}\right)^2 > \left(\frac{c_j}{c_i}\right)^{\eta+1},$$
(21)

then a small access markup increases α_i^* , but lowers the access revenues of network i. The impact on total industry profits is ambiguous as $\pi_1^H + \pi_2^H$ is increased, but the term $(\alpha_1^* - \alpha_2^*) (q_2^* - q_1^*) (a - c_o)$ is negative.

Proof. A formal proof is given in appendix 9.2. In order to show in more detail what drives the results of proposition 6, we give here a sketch of the proof. Profits are the sum of retail revenues

$$\alpha_i [t_i - c_i q_i - f_i] \equiv \alpha_i R_i$$

and access revenues A_i

$$A_i = \alpha_i (1 - \alpha_i)(q_i - q_i)(a - c_o)$$

A first difference with symmetric equilibria is that changes in market shares may affect the access revenue. As $A_1 = -A_2$ and $\alpha_1 = 1 - \alpha_2$

$$A' \equiv \frac{\partial A_1}{\partial \alpha_1} = \frac{\partial A_2}{\partial \alpha_2} = (\alpha_1 - \alpha_2)(q_1 - q_2)(a - c_o). \tag{22}$$

The cost to network i of a tariff cut needed to increase its market share by ε equals $\alpha_i \frac{\varepsilon}{\mu \sigma}$. The benefits are the extra retail profits, εR_i , plus the change in access revenues, $\varepsilon A'$. In equilibrium, retail profits per customer thus satisfy

$$R_1^* = \frac{\alpha_1^*}{\mu\sigma} - A'$$
 and $R_2^* = \frac{\alpha_2^*}{\mu\sigma} - A'$

The term A' reflects the incentives of networks to boost access revenues by increasing their market share. If networks can increase access revenues by having a higher market share (A' > 0), then competition for market share is tougher and retail profits will be lower. If access revenues are reduced by a higher market share (A' < 0), competition is softer and retail profits are higher. As flows between networks - and thus the absolute size of access deficits and revenues - increase when market shares are more equal $(\alpha_1\alpha_2)$ is maximal for $\alpha_1 = \alpha_2 = 0.5$, both the largest and the smallest network compete more fiercely for market share if the largest network faces an access deficit. Similarly, if the smallest network incurs an access deficit, an increase in the access charge gives both networks incentives to decrease their market share and thus to raise their tariff. These incentives to lower or increase tariffs in order to boost access revenues, however, do not affect the equilibrium market share, as the latter depends only on $R_2^* - R_1^*$,

$$lpha_2 = \mu \left(rac{1}{2} + \sigma \left[w_2 - w_1
ight] \right) = \mu \left(rac{1}{2} + \sigma \left[s_2(q_i) - R_2^* - s_1(q_j) + R_1^*
ight] \right)$$

Given $s_i(q_i)$ and $s_j(q_j)$, α_i^* is thus still identical as in the Hotelling model with unit demands, $\alpha_i^* = \alpha_i^H(s_i(q_i), s_j(q_j))$, but equilibrium profits now equal

$$\pi_i^* = \alpha_i R_i + A_i = \frac{\left(\alpha_i^H\right)^2}{\mu \sigma} - \alpha_i A' + A_i = \pi_i^H - \alpha_i A' + A_i$$
 (23)

yielding total industry profits of

$$\pi_i^H + \pi_j^H - A' \tag{24}$$

Suppose first that q_i^* and q_2^* are given or determined exogeneous. Total industry profits are then increased by an access markup if and only if the access charge softens competition for market share (A' < 0). Nevertheless, even when A' < 0, there is no scope for collusion: the absolute size of the access deficit incurred by one network $|A_j|$, is always larger than the gains from softer competition, $-\alpha_i A'$, induced by this access deficit.¹²

Let us now take the impact of a on q_i^* and thus on α_i and π_i^H into account. As usual, marginal prices are set at perceived marginal cost

$$q_i^* = q(p_i^*) = q(c_i + \alpha_j(a - c_o))$$
 (25)

$$-\alpha_j A' = -\alpha_j \frac{A_j}{\alpha_j} + \alpha_j \frac{A_j}{\alpha_i} = |A_j| - \frac{\alpha_j}{\alpha_i} |A_j| < |A_j|$$

so that profits of network j incurring an access deficit are

$$\pi_j^H - \frac{\alpha_j}{\alpha_i} |A_j|$$

¹²Suppose $A_j < 0$, then

Market shares depend on $s_i(q_i^*) - s_j(q_j^*)$, where $s_i(q_i) = u(q_i) - c_i q_i - f_i$ decreases with q_i for $q_i > q(c)$. From (25), if $c_i \approx c_j$ and $\alpha_i > \alpha_j$ for $a = c_o$, an access markup increases $q_i^* - q_j^*$ and thus $s_i(q_i^*) - s_j(q_j^*)$: if marginal costs are more or less equal, an access markup increases the largest market share. On the other hand, if $\alpha_i \approx \alpha_j$ for $a = c_o$, and $c_i < c_j$, a small access markup decreases $s_i(q_i^*) - s_j(q_j^*)$: if market shares are more or less equal, a small access markup lowers the market share of the network with the smallest marginal cost. We distinguish two cases:

- (1) If $c_i < c_i$ and $\alpha_i < \alpha_j$ for $a = c_o$, both forces go to decrease α_i . As network i then also incurs an access deficit, a small access markup lowers profits of network i. On the other hand, since a decrease in α_i increases $\pi_i^H + \pi_j^H$ and from (22), A' < 0, total industry profits, given by (24), are increased. This result goes also in the other direction. If for $a > c_o$, A' < 0, that is if an access markup softens competition, then the smallest network incurs an access deficit and, from (25), has a smaller marginal cost: if an access markup softens competition, then it also lowers profits of the smallest network.
- (2) If $c_i < c_i$ and $\alpha_i > \alpha_j$ for $a = c_o$, then an access markup results in an access deficit for the largest network (network i) and thus toughens competition for market share (A' > 0). Profits of network i are given by

$$\pi_i^* = \pi_i^H - \frac{\alpha_i}{\alpha_i} |A_i| < \pi_i^H - |A_i|$$

where $\pi_i^H = (\alpha_i)^2/\sigma$. If α_i/α_j is relatively small compared to c_j/c_i , then an access markup still decreases the market share of the network incurring an access deficit (network i) - and thus its profits. The exact condition is given by (20). From (24), also total industry profits are lowered. On the other hand, if α_i/α_j is large relative to c_j/c_i , an access markup increases α_i , and so that, a priori, its impact on π_i^* is ambiguous.

Proposition 6 tells us that networks' profits equal the standard Hotelling profits for unit demands, as in a symmetric equilibrium, plus, for $a \neq c_o$, a term specific to network competition. Moreover, the access charge may affect the standard Hotelling profits through its impact on $s_i(q_i) - s_j(q_j)$ (and thus on market shares), whereas the latter are independent of a in a symmetric equilibrium. From (18), the access charge essentially changes the level playing field between the two networks: neglecting its impact on market shares, an access markup lowers profits of the network which incurs an access deficit and raises those of the network incurring an access revenue. For small access markups, this is always the network with the lowest marginal cost (which we denote by network i). Taking the impact of an access markup on market shares - and thus on π_i^H - into account, lowers the profits of network i even more if it also has a smaller market share for $a = c_o$ or, in the opposite case, if α_i/α_j is relative small compared to c_j/c_i . This case is typical for entry by a low marginal cost operator. Due to its partial coverage of

the market or due to switching $costs^{13}$, the entrant tends to have a smaller market share, or, if $c_i \gg c_j$ only a slightly larger market share, than the incumbent. It is then in the interest of the incumbent to insist on a high access charge. While ex post, this leads to above marginal cost pricing (for $a > c_o, q_i^*$ and q_j^* are smaller than q(c)), ex ante, the prospect of high access charges may also deter entry by a low marginal cost entrant if there are substantial fixed investments and/or the low marginal cost technology involves a large fixed cost per customer.

If, on the other hand, α_i/α_j is relative large compared to c_j/c_i , then an access markup still results in an access deficit for network i, but it increases also its market share. A priori, the impact on profits is thus ambiguous. But, as we have shown above, the access charge then also increases competition for market share which lowers profits of both networks, as is reflected by the (negative) term $(a - c_o)(\alpha_1^* - \alpha_2^*)(q_2^* - q_1^*)$ in (19). Intuitively, the impact of the access deficit and this increased competition is likely to dominate the advantages of a possibly larger market share.

To put our analytical results in perspective, we report simulation results on competition between an entrant, E, with partial coverage ($\mu = 0.8$) and an incumbent, I, which may only differ in marginal costs.¹⁴

$$a = c_o \quad a - c_o = 0.2 \quad a - c_o = 0.4$$

$$c_E = 0.9 < c_I = 1 \quad \pi_I = \quad .146 \quad .134 \quad .126$$

$$\alpha_I = \quad .459 \quad .459 \quad .459 \quad .459$$

$$c_E = 1 > c_I = 0.9 \quad \pi_I = \quad .184 \quad .168 \quad .151$$

$$\alpha_I = \quad .607 \quad .611 \quad .618$$

$$\pi_E = \quad .1089 \quad .1091 \quad .1097$$

$$c_E = c_I = 1 \quad \pi_I = \quad .1422 \quad .1418 \quad .1407$$

$$\alpha_I = \quad .533 \quad .535 \quad .538$$

$$\alpha_2 = \frac{1}{2} - \frac{S}{2} + \sigma \left[w_2 - w_1 \right]$$

such that

$$\alpha_2^* = \alpha_2^H = \frac{1}{2} - \frac{S}{6} + \frac{\sigma}{3} \left[s(q_1^*) - s(q_2^*) \right].$$

 $^{^{13}}$ Proposition 6 is robust to the case where customers incur a fixed switching cost S when they join the entrant. If the entrant has full coverage, its market share is then

¹⁴Other parameter values are $\eta = 2$, $\sigma = 2.5$ and $f_i = f_j = 0$. Simulations where $\mu = 1$, but $f_i < f_j$ would yield similar results: from the expression for α_i^H , differences in fixed costs have the same impact on market shares - and thus on profits - as differences in coverage.

- The first case deals with entry by a low marginal cost operator $(c_E < c_I)$. Since condition (21) of proposition 6 holds practically at the equality, the access markup has almost no impact on the market shares. Profits, on the other hand, are considerably affected: the entrant's profits fall by $\pm 15\%$ while the incumbent's profits increase by $\pm 15\%$.
- In the second case, the incumbent has a smaller marginal cost $(c_E < c_I)$, and condition (21) is largely satisfied. As predicted, the access markup increases the market share of the incumbent. Nevertheless, the incumbent's profits fall by more than 15%. Also total industry profits falls by some 8%, as the access deficit induces tougher competition for market share. The impact of the access charge on the access deficit and on competition thus by far outweighs the benefits of a larger market share.
- Finally, we look at the case where marginal costs are identical, which is not covered by proposition 6. The impact on profits is then negligeable (less than 0.1% of profits).

8 Concluding remarks

Previous research on network interconnection and two-way access has mainly focussed on linear pricing, thereby emphasizing the collusive power of the access charge. Given that competition between telecommunications operators de facto is competition in nonlinear pricing, this can only be justified by an implicit assumption that linear prices are a good short cut of nonlinear prices. Though the seminal work of Laffont et al. has shown that in a simple model with homogeneous customers and a symmetric equilibrium, the collusive effect of the access charge then completely disappears, both Laffont et al. (1998a,b) and Armstrong (1998) argued that once customers are heterogeneous in demand and marginal prices differ from marginal cost, results are likely to resemble those under linear pricing more closely. Since subsequent research and policy oriented publications have continued to report results obtained under the assumption of linear pricing, often without mentioning the result on two-part tariffs in Laffont et al., the idea that the access charge is an instrument of collusion has become widespread among policy makers.¹⁵

This paper has argued that the collusive effect of the access charge totally depends on the - in reality not satisfied - assumption of competition in linear pricing. Our results suggest that once sufficient (network) competition has been developed, regulation of both final prices and access charges can be withdrawn, that is the telecommunication industry is potentially competitive. This, however, is not necessarily true in the entry phase of competition when networks tend to differ in their cost structure and/or coverage. The access charge then has a level-

 $^{^{15}}$ See e.g. Armstrong (1997), Carter and Wright (1997), Carter and Wright (1999), Doganoglu and Tauman (1996).

playing field effect: it increases profits of one network, and lowers profits of the other. If networks have different marginal costs, this impact on profits may be huge. To the extent that the bargaining power is unequally distributed among networks, the negotiated access charge is likely to differ from the marginal cost of access. A regulatory body which intervenes in case operators do not reach an agreement may then be necessary to restore efficiency. Future research on two-way access should investigate more thoroughly how a high (or low) access charges may give an unfair competitive advantage to a particular network¹⁶ and examine further the (determinants of the) bargaining process.¹⁷ The regulatory framework, that is wether or not access charges must be reciprocal, if there is a mandated access charge in case of no agreement and so on, is likely to affect considerably the efficiency of the bargaining outcome.

References

- [1] Armstrong M. (1997) Competition in Telecommunications. Oxford Review of Economic Policy. Vol. 13, No.1, pp.64-81.
- [2] Armstrong, M. (1998). Network Interconnection in telecommunications. *The Economic Journal*, Vol 108, pp.545-564.
- [3] Armstrong, M., and J. Vickers (1998) Competitive Price Discrimination, mimeo.
- [4] Carter, M., and J. Wright. (1997) Bargaining over interconnection: the Clear-Telecom dispute. Mimeo, University of Auckland.
- [5] Carter, M., and J. Wright (1999) Interconnection in network industries. Review of Industrial Organization, forthcoming.
- [6] Dessein, W. (1999) Network Competition with Heterogeneous Calling Patterns, mimeo.
- [7] Doganoglu, T. and Tauman, Y. (1996). "Network Competition with Reciprocal Proportional Access Charge Rules." Mimeo, SUNY at Stony Brook and Tel-Aviv University.
- [8] Laffont, J.J., Rey, P. and J. Tirole (1997). Competition between Telecommunications Operators. *European Economic Review*, Vol.41(3-5), pp. 701-711

¹⁶For example, our companion paper Dessein (1999) shows how under certain conditions, the incumbent, by setting an appropriate access charge, can impede entry on a segment wich tends to have a net outflow of calls.

¹⁷See e.g. Carter and Wright (1997) which analyzes explicitly the bargaining over interconnection, but in case of linear pricing.

- [9] Laffont, J.J., Rey, P. and J. Tirole (1998a). Network Competition: I. Overview and Nondiscriminatory Pricing. Rand Journal, Vol.29, No.1, pp.1-37
- [10] Laffont, J.J., Rey, P. and J. Tirole (1998b). Network Competition: II. Price Discrimination. *Rand Journal*, Vol.29, No.1, pp.38-56.
- [11] Laffont, J.J., and J. Tirole (1996). Creating Competition through Interconnection: Theory and Practice. *Journal of Regulatory Economics*, 10, pp 227-256.
- [12] Laffont, J.J., and J. Tirole (1999). Competition in Telecommunications. Cambridge: MIT Press, forthcoming.
- [13] Rochet, J.-C. and L. Stole (1998), 'Selection Contracts with Uncertain Participation: Re-examining Monopoly and Competitive Nonlinear Pricing', mimeo, Toulouse.
- [14] Tirole, J. (1988). The Theory of Industrial Organisation. Cambridge: MIT Press
- [15] Varian, H. (1989), 'Price Discrimination', in R. Schmalensee and R. Willig, eds., *Handbook of Industrial Organization*, *Volume 1*. Amsterdam: North Holland.
- [16] Wilson, R.(1992). Nonlinear Pricing. Oxford: Oxford University Press

9 Appendix

9.1 Implicit price discrimination.

9.1.1 Standard model

Proof of lemma 2:

As we make also use of lemma 2 in the proof of proposition 4, we provide a proof for the more general case in which σ_H may differ from σ_H , although we restrict ourselves to the case in which $a>c_o$, the case $a< c_o$ being similar. As symmetric equilibria in which networks serve only one segment can easily be ruled out for k_L close to k_H , the first order conditions must be satisfied. For $\lambda_L=\lambda_H=0$, it follows from the FOC that the symmetric equilibrium is uniquely defined and given by $\{\hat{q}_L,\hat{t}_L,\hat{q}_H,\hat{t}_H\}$. Consequently, if $\{\hat{q}_L,\hat{t}_L,\hat{q}_H,\hat{t}_H\}$ is not incentive compatible and if a symmetric equilibrium exists, at least one incentive constraint is strictly binding: $\lambda_H>0$ and/or $\lambda_L>0$. We characterize

the incentive compatible symmetric equilibrium contract, $\{t_L^*, q_L^*, t_H^*, q_H^*\}$. The first order condition with respect to q_L and q_H yield

$$\mu \alpha_L \left[k_L^{1/\eta} q_L^{*-1/\eta} - \left(c + \frac{a - c_o}{2} \right) \right] - \lambda_H \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_L^{*-1/\eta} = 0 \quad (26)$$

$$(1 - \mu) \alpha_H \left[k_H^{1/\eta} q_H^{*-1/\eta} - \left(c + \frac{a - c_o}{2} \right) \right] + \lambda_L \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_H^{*-1/\eta} = 0 \quad (27)$$

with λ_H and/or λ_L strictly positive. Incentive constraints can never be binding at the same time: setting both IC's at equality and subtracting yields $q_H^* = q_L^* = q^*$; however, from (26) and (27), one must have then that

$$\left[k_H^{1/\eta} - k_L^{1/\eta}\right] (q^*)^{-1/\eta} + \left(\frac{\lambda_L}{(1-\mu)\alpha_H} + \frac{\lambda_H}{\mu\alpha_L}\right) \left[k_H^{1/\eta} - k_L^{1/\eta}\right] (q^*)^{-1/\eta} = 0$$

which is impossible if both λ_L and λ_H are positive. Thus either $(\lambda_H > 0, \lambda_L = 0)$ or $(\lambda_H = 0, \lambda_L > 0)$. The proof can now be constructed by contradiction: $(\lambda_H = 0, \lambda_L > 0)$ holds if $(\lambda_H > 0, \lambda_L = 0)$ is impossible and the other way round. We distinguish two case:

a) The incentive constraint of the light users is violated in the explicit price discrimination equilibrium.

Suppose that $\{w_L^*, q_L^*, w_H^*, q_H^*\}$ is then such that the IC of the heavy users is binding, thus $\lambda_H > 0$ and $\lambda_L = 0$. We then have

$$w_H^* - w_L^* = \frac{\eta}{\eta - 1} \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_L^{*1 - 1/\eta}$$
 (28)

while from the FOC with respect to w_L and w_H ,

$$\left[\frac{\eta}{\eta - 1} k_L^{1/\eta} q_L^{*1 - 1/\eta} - w_L^* - c q_L^* - f\right] + \frac{(a - c_o)}{\mu} S = \lambda_H / \mu \sigma_L + 1/2\sigma_L \tag{29}$$

$$\left[\frac{\eta}{\eta - 1} k_H^{1/\eta} q_H^{*1 - 1/\eta} - w_H^* - c q_H^* - f\right] - \frac{(a - c_o)}{1 - \mu} S = -\lambda_H / (1 - \mu) \sigma_H + 1/2 \sigma_H$$
 (30)

where S is the net inflow of calls in the light user's segment:

$$S = S(\lambda_L, \lambda_H) \equiv (\ell [\mu q_L^* + (1 - \mu) q_H^*] - \mu q_L^*)$$

Subtracting (29) and (30) and substituting (28), we find

$$\frac{\eta}{n-1}k_H^{1/\eta}\left[q_H^{*1-1/\eta} - q_L^{*1-1/\eta}\right] - c(q_H^* - q_L^*) \tag{31}$$

$$=\frac{\sigma_L-\sigma_H}{2\sigma_H\sigma_L}+(a-c_o)\frac{S(\lambda_L,\lambda_H)}{\mu(1-\mu)}-\lambda_H/(1-\mu)\sigma_H-\lambda_H/\mu\sigma_L$$

For $\delta = k_H - k_L$ small, $S(\lambda_L, \lambda_H)$ can be approximated by a Taylor expansion. As for $\lambda_H > 0$ and $\lambda_L = 0$,

$$q_H^* = \hat{q}_H \quad \text{and} \quad q_L^* = \hat{q}_L \left(1 - \frac{2\lambda_H}{\mu} \frac{\left[k_H^{1/\eta} - k_L^{1/\eta} \right]}{k_L^{1/\eta}} \right)^{\eta},$$

we find after some computations that

$$S(\lambda_L, \lambda_H) \cong \delta\left(\frac{\partial S(\lambda_L, \lambda_H)}{\partial \delta}_{|\delta=0, \lambda_H=\lambda_H^*}\right) = 2\delta\lambda_H (1-\mu)(c + \frac{a-c_o}{2})^{-\eta}$$

It follows that for δ or $a-c_o$ small, $\lambda_H/(1-\mu)\sigma_H + \lambda_H/\mu\sigma_L - (a-c_o)\frac{S(\lambda_L,\lambda_H)}{\mu(1-\mu)} > 0$, so that

$$\frac{\eta}{n-1}k_H^{1/\eta} \left[q_H^{* \ 1-1/\eta} - q_L^{* \ 1-1/\eta} \right] - c(q_H^* - q_L^*) < \frac{\sigma_L - \sigma_H}{2\sigma_H \sigma_I}$$
 (32)

On the other hand, the explicit price discrimination outcome $\{\hat{q}_L, \hat{w}_L, \hat{q}_H, \hat{w}_H\}$ satisfies¹⁸

$$\left[\frac{\eta}{\eta - 1} k_L^{1/\eta} \hat{q}_L^{1 - 1/\eta} - \hat{w}_L - c\hat{q}_L - f\right] = 1/2\sigma_L \tag{33}$$

$$\left[\frac{\eta}{\eta - 1} k_H^{1/\eta} \hat{q}_H^{1 - 1/\eta} - \hat{w}_H - c\hat{q}_H - f\right] = 1/2\sigma_H,\tag{34}$$

and

$$\hat{w}_H - \hat{w}_L > \frac{\eta}{\eta - 1} \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_L^{1 - 1/\eta} \tag{35}$$

Subtracting (33) and (34) and taking (35) and $q_H^* = \hat{q}_H$ into account, we find

$$\frac{\eta}{n-1}k_H^{1/\eta} \left[q_H^{*1-1/\eta} - \hat{q}_L^{1-1/\eta} \right] - c(q_H^* - \hat{q}_L) > \frac{\sigma_L - \sigma_H}{2\sigma_H \sigma_L}$$
 (36)

But

$$\frac{\partial \left(\frac{\eta}{\eta - 1} k_H^{1/\eta} q^{1 - 1/\eta} - c q_L\right)}{\partial q} = k_H^{1/\eta} q^{-1/\eta} - c > 0 \Leftrightarrow q < k_H q(c)$$

As for $\lambda_H > 0$, $q_L^* < \hat{q}_L = k_L q(c + \frac{a - c_o}{2}) < k_H q(c)$, from (36), also

$$\frac{\eta}{n-1}k_H^{1/\eta} \left[q_H^{*1-1/\eta} - q_L^{*1-1/\eta} \right] - c(q_H^* - q_L^*) > \frac{\sigma_L - \sigma_H}{2\sigma_H \sigma_L}$$
 (37)

which is in contradiction with (31): $\lambda_H > 0$ and $\lambda_L = 0$ is impossible.

b) The incentive constraint of the heavy users is violated in the explicit price discrimination equilibrium.

Suppose that $\{w_L^*, q_L^*, w_H^*, q_H^*\}$ is then such that the IC of the light users is binding, thus $\lambda_L > 0$ and $\lambda_H = 0$. We then have

$$\begin{split} w_H^* - w_L^* &= \frac{\eta}{\eta - 1} \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_H^{* \ 1 - 1/\eta} \\ & \left[\frac{\eta}{\eta - 1} k_L^{1/\eta} q_L^{* \ 1 - 1/\eta} - w_L^* - c q_L^* - f \right] + \frac{(a - c_o)}{\mu} S = -\lambda_L / \mu \sigma_L + 1/2 \sigma_L \\ & \left[\frac{\eta}{\eta - 1} k_H^{1/\eta} q_H^{* \ 1 - 1/\eta} - w_H^* - c q_H^* - f \right] - \frac{(a - c_o)}{1 - \mu} S = \lambda_L / (1 - \mu) \sigma_H + 1/2 \sigma_H, \end{split}$$

From our balanced calling pattern assumption, $A_L(\hat{q}_L, \hat{q}_H) = A_L(\hat{q}_L, \hat{q}_H) = 0$.

from which

$$\begin{split} &\frac{\eta}{\eta - 1} k_L^{1/\eta} \left[q_H^{* \ 1 - 1/\eta} - q_L^{* \ 1 - 1/\eta} \right] - c (q_H^* - q_L^*) \\ &= \frac{\sigma_L - \sigma_H}{2\sigma_H \sigma_L} + \frac{(a - c_o)}{\mu (1 - \mu)} S + \lambda_L / (1 - \mu) \sigma_H + \lambda_L / \mu \sigma_L \end{split}$$

On the other hand, the explicit price discrimination outcome $\{\hat{q}_L, \hat{w}_L, \hat{q}_H, \hat{w}_H\}$ satisfies now

$$\frac{\eta}{\eta - 1} k_L^{1/\eta} \left[\hat{q}_H^{1 - 1/\eta} - \hat{q}_L^{1 - 1/\eta} \right] - c(\hat{q}_H - \hat{q}_L) < \frac{\sigma_L - \sigma_H}{2\sigma_H \sigma_L}$$

Subtracting the previous equations, we find:

$$\frac{\eta}{\eta - 1} k_L^{1/\eta} \left[q_H^{* \, 1 - 1/\eta} - \hat{q}_H^{1 - 1/\eta} \right] - c(q_H^* - \hat{q}_H) > \frac{\lambda_L}{(1 - \mu)\sigma_H} + \frac{\lambda_L}{\mu \sigma_L} + \frac{(a - c_o)}{\mu (1 - \mu)} S$$
(38)

For $\delta = k_H - k_L$ small, the LHS of (38) can be approximated by a Taylor expansion. As for $\lambda_L > 0$ and $\lambda_H = 0$,

$$q_L^* = \hat{q}_L \quad ext{and} \quad q_H^* = \hat{q}_H \left(1 + rac{2\lambda_L}{(1-\mu)} rac{\left[k_H^{1/\eta} - k_L^{1/\eta}
ight]}{k_H^{1/\eta}}
ight)^{\eta},$$

we find

$$\frac{\eta}{\eta - 1} k_L^{1/\eta} \left[q_H^{* \ 1 - 1/\eta} - \hat{q}_H^{1 - 1/\eta} \right] - c(q_H^* - \hat{q}_H) \cong \delta \frac{\lambda_L}{(1 - \mu)} (c + \frac{a - c_o}{2})^{-\eta} (a - c_o)$$

On the other hand, as $q_L^* = \hat{q}_L$ while $q_H^* > \hat{q}_H$, we have S > 0, such that

$$\lambda_L/(1-\mu)\sigma_H + \lambda_L/\mu\sigma_L + \frac{(a-c_o)}{\mu(1-\mu)}S > \lambda_L/(1-\mu)\sigma_H + \lambda_L/\mu\sigma_L$$

It follows that for δ small, (38) is violated: $\lambda_L > 0$ and $\lambda_H = 0$ is impossible. \blacksquare Proof of proposition 2.

1) Equilibrium profits are independent of the access charge:

We first show that if a symmetric equilibrium exists, profits per firm are independent of a and equal $1/4\sigma$. We maximize the network's profits, given by (2), subject to the incentive constraints, which can be rewritten as

$$w_H - w_L \ge \frac{\eta}{\eta - 1} \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_L^{1 - 1/\eta}$$
 (39)

$$w_H - w_L \le \frac{\eta}{\eta - 1} \left[k_H^{1/\eta} - k_L^{1/\eta} \right] q_H^{1 - 1/\eta}$$
 (40)

With λ_H and λ_L the lagrange multipliers of the incentive constraint for respectively the heavy users and the light users, the FOC with respect to w_L and w_H yield

$$\lambda_{H} - \lambda_{L} + \mu/2 = \mu \sigma \left[\frac{\eta}{\eta - 1} k_{L}^{1/\eta} q_{L}^{1 - 1/\eta} - w_{L} - cq_{L} - f \right] - \sigma \left(\mu q_{L} - \ell \left[\mu q_{L} + (1 - \mu) q_{H} \right] \right) (a - c_{o})$$
(41)

$$-\lambda_H + \lambda_L + (1 - \mu)/2 = (1 - \mu)\sigma \left[\frac{\eta}{\eta - 1} k_H^{1/\eta} q_H^{1 - 1/\eta} - w_H - cq_H - f \right] + \sigma \left(\mu q_L - \ell \left[\mu q_L + (1 - \mu) q_H \right] \right) (a - c_o)$$
(42)

Summing up and substituting w_L and w_H , we find

$$\mu [t_L - cq_L - f] + (1 - \mu) [t_H - cq_H - f] = 1/2\sigma$$

But in a symmetric equilibrium, each operator gets exactly

$$1/2 \left[\mu \left(t_L - cq_L - f \right) + (1 - \mu) \left(t_H - cq_H - f \right) \right]$$

Hence, equilibrium profits per firm are $1/4\sigma$.

2) Equilibrium contract:

For $a=c_o$, network i' s profits on the customer segment s are strictly concave in $\{q_s,w_s\}$. As a result, total profits given k_L,k_H are strictly concave in $\{q_L,w_L,q_H,w_H\}$ for $a=c_o$ and the Hessian matrix $D^2\pi(q_L,w_L,q_H,w_H)$ is negative semidefinite for $a=c_o$. Fix the average customer type k. As all terms of $D^2\pi(q_L,w_L,q_H,w_H)$ are continuous in k_L,k_H and a, then for any $\delta'=k_H-k_L$, one can find an access charge $a'>c_o$ such that $D^2\pi(q_L,w_L,q_H,w_H)$ is still negative semidefinite and thus profits are strictly concave, for $c_o\leq a\leq a'$ or $0\leq \delta\leq \delta'$. A candidate equilibrium satisfying the FOC is then effectively an equilibrium. From these FOC, if networks can discriminate explicitly, a unique symmetric equilibrium exists and is given by $\{\hat{q}_L,\hat{u}_L,\hat{q}_H,\hat{u}_H\}$ (cf. section 4). As shown in the text, given $k_H-k_L\in]0,\delta']$, for a close enough to c_o , IC's are satisfied by $\{\hat{q}_L,\hat{u}_L,\hat{q}_H,\hat{u}_H\}$, which is thus also the equilibrium under implicit price discrimination. On the other hand, given $a\in]c_o,a']$, for k_L close to k_H , the IC of the light users is violated by $\{\hat{q}_L,\hat{u}_L,\hat{q}_H,\hat{u}_H\}$. From lemma 2, then $\lambda_L>0$ and $\lambda_H=0$ and symmetric equilibrium quantities are characterized by

$$q_L^* = \hat{q}_L = k_L q(c + \frac{a - c_o}{2})$$
 and $q_H^* = \hat{q}_H \left(1 + \frac{2\lambda_L}{(1 - \mu)} \frac{\left[k_H^{1/\eta} - k_L^{1/\eta}\right]}{k_H^{1/\eta}}\right)^{\eta} > \hat{q}_H$

Similarly from (41) and (42), profits per heavy user, π_H^* , and light user, π_L^* , are

$$\pi_H^* = \frac{1}{2\sigma} + \frac{\lambda_L}{\sigma(1-\mu)}$$
 and $\pi_L^* = \frac{1}{2\sigma} - \frac{\lambda_L}{\sigma(\mu)}$

and thus

$$t_L^* = \hat{t}_L - \frac{\lambda_L}{\sigma \mu} < \hat{t}_L \quad and \quad t_H^* - cq_H^* = \hat{t}_H - c\hat{q}_H + \frac{\lambda_L}{\sigma(1-\mu)} > \hat{t}_H - c\hat{q}_H.$$

¹⁹See Laffont-Rev-Tirole 1998a, appendix B.

9.1.2 Two-part tariffs

Proof of proposition 3:

Equilibrium profits: Competition can be seen as one in which networks compete in net surpluses (w_L, w_H) and usage fees rather than fixed fees and usage fees, in which case profits are given by (2) with $q_H = k_H q(p_H)$ and $q_L = k_L q(p_L)$. Incentive constraints can be rewritten as

$$w_H - w_L \ge (k_H - k_L)v(p_L) \tag{43}$$

$$w_H - w_L \leq (k_H - k_L)v(p_H) \tag{44}$$

The proof that in any symmetric equilibrium, profits are independent of the access charge and equal $1/4\sigma$, is now identical to the one in the standard model.

Equilibrium contract: Profits are strictly concave for $a=c_o$ and thus also for a close to c_o . As argued in the text, for $a>c_o$, the equilibrium contract under explicit price discrimination is not implementable in two-part tariffs as then the IC's of the light users would be violated. As the proof of lemma 2 can be applied for competition in two-part tariffs, this IC is then binding in equilibrium. From the FOC with respect to p_L and p_H :

$$p_L^* = c + \frac{a - c_o}{2}$$
 and $p_H^* = \frac{c + \frac{a - c_o}{2}}{1 + \frac{2\lambda_L}{(1 - \mu)} \frac{\left[k_H^{1/\eta} - k_L^{1/\eta}\right]}{k_H^{1/\eta}}} < p_L^*$

From the FOC with respect to w_L and w_H , profits per heavy user, π_H^* , and light user, π_L^* , are

$$\pi_H^* = \frac{1}{2\sigma} + \frac{\lambda_L}{\sigma(1-\mu)}$$
 and $\pi_L^* = \frac{1}{2\sigma} - \frac{\lambda_L}{\sigma\mu}$

and thus

$$F_{L}^{*} = 1/2\sigma + f - \frac{a - c_{o}}{2}k_{L}q(p_{L}) - \frac{\lambda_{L}}{\sigma\mu}$$

$$F_{H}^{*} = 1/2\sigma + f - (p_{H} - c)k_{H}q(p_{H}) + \frac{\lambda_{L}}{\sigma(1 - \mu)}$$

Of course, as $p_L^* > p_H^*$, one must also have $F_H^* > F_L^*$.

9.1.3 Limits to the neutrality of the access charge on profits

Proof of proposition 4.

Proof. From the FOC with respect to (w_L, q_L) and (w_H, q_H) , the equilibrium under explicit price discrimination is characterized by (6) and (7) and (8) and profits under explicit price discrimination are $\pi_D^* = \mu/4\sigma_L + (1-\mu)(4\sigma_H)$. From the first order condition with respect to w_L and w_H , profits under implicit price

discrimination are given by $\pi^* = \pi_D^* + (\lambda_H - \lambda_L) (1/2\sigma_L - 1/2\sigma_H)$ with λ_H and λ_L the lagrange multipliers of the IC of respectively heavy users and light users. This proves (11) and (10). From (6) and (7) and (8), the explicit price discrimination equilibrium satisfies the IC of light users and heavy if and only if respectively (12) and (13) are satisfied.

We are now ready to prove the first part of the proposition. As for $a = c_o$, $V_H(\hat{q}_L) - V_H(\hat{q}_H)$ is negative and $V_L(\hat{q}_L) - V_L(\hat{q}_H)$ is positive, for $|\sigma_H - \sigma_L|$ sufficiently small $|F_H - F_L|$ tends to zero and both IC's are satisfied and $\pi^* = \pi_D^*$. On the other hand, both $V_L(\hat{q}_L) - V_L(\hat{q}_H)$ and $V_H(\hat{q}_L) - V_H(\hat{q}_H)$ go to zero as Δk goes to zero, such that given $|\sigma_H - \sigma_L|$, the IC of customers with the smallest perceived substitutability are violated for Δk sufficiently small. From lemma 2, the same IC then is binding in the equilibrium under implicit discrimination and from (11) and (10), $\pi^* < \pi_D^*$.

The proof of the second part goes as follows. (1) When $c + \frac{a-c_o}{2}$ goes to zero, $\hat{q}_H - \hat{q}_L$ tends to infinity such that $V_s(\hat{q}_L) - V_s(\hat{q}_H)$, (s = L, H) increases without a bound when a gets smaller. For a small enough IC_L is then always satisfied, while IC_H will be violated. From lemma 2, for Δk small, the IC of the heavy users is then binding under implicit discrimination such that from (11) and (10), $\pi^* > \pi_D^*$ if $\sigma_H > \sigma_L$ and $\pi^* < \pi_D^*$ if $\sigma_L < \sigma_H$. (2) A negative access markup $(a < c_o)$ always decreases $V_H(\hat{q}_L) - V_H(\hat{q}_H)$ and $V_L(\hat{q}_L) - V_L(\hat{q}_H)$, which reach a minimum respectively for $a_H^* > c_o$ and $a_L^* > c_o$ where

$$1 + (a_H^* - c_o)/2c = \Delta k / \left(k_H - k_H^{1/\eta} k_L^{1-1/\eta}\right)$$

$$1 + (a_H^* - c_o)/2c = \Delta k / \left(k_H^{1-1/\eta} k_L^{1/\eta} - k_L\right).$$

Denote this minimum respectively by $-\bar{V}_H$ and $-\bar{V}_L$. As for $a \geq c_o$, $V_H(\hat{q}_L) - V_H(\hat{q}_H)$ is always negative and for $\hat{q}_H \leq k_L q(c)$, $V_L(\hat{q}_L) - V_L(\hat{q}_H)$ is always negative, both $-\bar{V}_H$ and $-\bar{V}_L$ are negative. For $|\sigma_H - \sigma_L|$ - and thus also $|F_H - F_L|$ - small enough, there exists then an $a' > c_o$, such that the IC of the light users is violated by the explicit price discrimination equilibrium. If also Δk is small, the IC of the light users is then binding under implicit discrimination and $\pi^* > \pi_D^*$ if $\sigma_H < \sigma_L$, $\pi^* < \pi_D^*$ if $\sigma_L > \sigma_H$.

9.2 Entry and asymmetric equilibria

Proof of proposition 6: Network *i* solves:

$$\max_{w_i, q_i} \{ \pi_i = \alpha_i(w_i, w_j) \left[s(q_i) - w_i \right] + A_i \}$$

with

$$s(q_i) = u(q_i) - cq_i - f_i$$

 $A_i = \alpha_i(w_i, w_i)(1 - \alpha_i(w_i, w_i))(q_i - q_i)(a - c_o)$

Equilibrium profits: In any shared market equilibrium, the first order condition with respect to w_i must be satisfied, this yields:

$$\mu \sigma \frac{\pi_i}{\alpha_i} - \alpha_i - \mu \sigma \alpha_i (q_j - q_i)(a - c_o) = 0$$
(45)

and thus

$$\pi_i = \frac{\alpha_i^2}{\mu \sigma} + \alpha_i^2 (q_j - q_i)(a - c_o) \equiv \alpha_i R_i + A_i$$
(46)

where

$$R_i = \frac{\alpha_i}{\mu\sigma} + (\alpha_i - \alpha_j)(q_j - q_i)(a - c_o)$$

Industry profits:

$$\pi_i + \pi_j = \alpha_i R_j + \alpha_i R_j = \frac{(\alpha_i)^2}{\mu \sigma} + \frac{(\alpha_j)^2}{\mu \sigma} + (\alpha_i - \alpha_j)(q_j - q_i)(a - c_o)$$

Equilibrium quantities: The first order condition with respect to q_i , yields

$$q_i^* = q(p_i^*) = q(c_i + \alpha_i(a - c_o)) \tag{47}$$

Equilibrium market shares: Suppose network 1 to be the incumbent in case of entry, α_1 is then given by

$$\alpha_{1} = 1 - \mu \frac{1}{2} + \mu \sigma(w_{1} - w_{2})$$

$$= 1 - \mu \frac{1}{2} + \mu \sigma \left[s(q_{1}) - R_{1} - s(q_{2}) + R_{2} \right]$$

$$= 1 - \mu \frac{1}{2} + \mu \sigma \left[s(q_{1}) - \frac{\alpha_{1}}{\mu \sigma} - s(q_{2}) + \frac{(1 - \alpha_{1})}{\mu \sigma} \right]$$

which yields

$$\alpha_1^* = \frac{1}{2} + \frac{1-\mu}{3} + \frac{\mu\sigma}{3} [s_1 - s_2]$$

Comparative statics: The comparative static results are a direct consequence of the following observations:

(1) Impact of a on market shares: We have

$$\frac{d\alpha_i^*}{da} = \frac{\mu\sigma}{3} \left[\frac{ds_i}{da} - \frac{ds_j}{da} \right]$$

As for $a = c_o$, $p_i^* = c_i$, and $\frac{ds_i(q(p_i))}{dp} | p_i = c_i = 0$, it follows that $\frac{d\alpha_i^*}{da} | a = c_o = 0$. A small access markup increases $s_i - s_i$ and thus α_i if

$$\frac{d^2s_j(q(p_j))}{da^2} \mid a = c_o < \frac{d^2s_i(q(p_i))}{da^2} \mid a = c_o \Leftrightarrow$$

$$\frac{\partial^2 s_j(q(p_j))}{\partial p^2} \left(\frac{dp_j}{da}\right)^2 \mid a = c_o < \frac{\partial^2 s_i(q(p_i))}{\partial p^2} \frac{dp_i}{da} \mid a = c_o$$
 where $\frac{dp_i}{da} \mid a = c_o = \frac{\partial p_i}{\partial a} \mid a = c_o = \alpha_j$, and

$$rac{\partial^2 s_i(q(p_i))}{\partial^2 p}\Big|_{a=c_o} = rac{\partial^2 \left(rac{p^{1-\eta}}{1-rac{1}{\eta}}-c_ip^{-\eta}
ight)}{\partial p^2}\Big|_{a=c_o} = -rac{\eta}{c_i^{\eta+1}},$$

A small access markup thus increases α_i if $(\alpha_i/\alpha_j)^2 > (c_j/c_i)^{\eta+1}$. Similarly, a small access markup decreases α_i if $(\alpha_i/\alpha_j)^2 < (c_j/c_i)^{\eta+1}$.

- (2) Impact of a on the access deficit: As $c_i < c_j$, $q_j^* > q_i^*$ for a small (but positive) and a small access markup results in an access deficit for network i. $(A_i < 0)$.
- (3) Impact of a on total industry profits: A sufficient condition for total industry profits to increase is that a small access markup decreases the smallest market share (as this increases $\alpha_i^2 + \alpha_j^2$), and results in an access deficit for the network with the smallest market share (as then $(\alpha_i \alpha_j)(q_j q_i)(a c_o) > 0$). A sufficient condition for total industry profits to decrease is that a small access markup decreases the largest market share and results in an access deficit for the network with the largest market share.

Bücher des Forschungsschwerpunkts Marktprozeß und Unternehmensentwicklung

Books of the Research Area Market Processes and Corporate Development

(nur im Buchhandel erhältlich/available through bookstores)

Tobias Miarka

Financial Intermediation and Deregulation: A Critical Analysis of Japanese Bank-Firm-Relationships

2000, Physica-Verlag

Damien J. Neven, Lars-Hendrik Röller (Eds.)

The Political Economy of Industrial Policy in

Europe and the Member States

2000, edition sigma

Jianping Yang

Bankbeziehungen deutscher Unternehmen: Investitionsverhalten und Risikoanalyse

2000, Deutscher Universitäts-Verlag

Horst Albach, Ulrike Görtzen, Rita Zobel Eds.)
Information Processing as a Competitive
Advantage of Japanese Firms

1999, edition sigma

Dieter Köster

Christian Wey

Wettbewerb in Netzproduktmärkten 1999, Deutscher Universitäts-Verlag

Marktorganisation durch Standardisierung: Ein Beitrag zur Neuen Institutionenökonomik des Marktes

1999, edition sigma

Horst Albach, Meinolf Dierkes, Ariane Berthoin Antal, Kristina Vaillant (Hg.)

Organisationslernen – institutionelle und kulturelle Dimensionen

WZB-Jahrbuch 1998 1998, edition sigma

Lars Bergman, Chris Doyle, Jordi Gual, Lars Hultkrantz, Damien Neven, Lars-Hendrik Röller, Leonard Waverman

Europe's Network Industries: Conflicting Priorities - Telecommunications

Monitoring European Deregulation 1 1998, Centre for Economic Policy Research

Manfred Fleischer

The Inefficiency Trap
Strategy Failure in the
German Machine Tool Industry
1997, edition sigma

Christian Göseke

Information Gathering and Dissemination

The Contribution of JETRO to Japanese Competitiveness 1997, Deutscher Universitäts-Verlag

Andreas Schmidt

Flugzeughersteller zwischen globalem Wettbewerb und internationaler Kooperation Der Einfluß von Organisationsstrukturen auf die Wettbewerbsfähigkeit von Hochtechnologie-Unternehmen

1997, edition sigma

Horst Albach, Jim Y. Jin, Christoph Schenk (Eds.) Collusion through Information Sharing?
New Trends in Competition Policy
1996, edition sigma

Stefan O. Georg

Die Leistungsfähigkeit japanischer Banken Eine Strukturanalyse des Bankensystems in Japan

1996, edition sigma

Stephanie Rosenkranz

Cooperation for Product Innovation

1996, edition sigma

Horst Albach, Stephanie Rosenkranz (Eds.) Intellectual Property Rights and Global Competition - Towards a New Synthesis 1995, edition sigma.

David B. Audretsch Innovation and Industry Evolution 1995, The MIT Press.

Julie Ann Elston

US Tax Reform and Investment: Reality and Rhetoric in the 1980s

1995, Avebury

Horst Albach

The Transformation of Firms and Markets: A Network Approach to Economic Transformation Processes in East Germany Acta Universitatis Upsaliensis, Studia Oeconomiae Negotiorum, Vol. 34 1994, Almqvist & Wiksell International (Stockholm).

DISCUSSION PAPERS 1999

Suchan Chae

Bargaining Power of a Coalition in Parallel Bargaining: FS IV 99 - 1

Paul Heidhues	Advantage of Multiple Cable System Operators	1010 33 1
Christian Wey	Compatibility Investments in Duopoly with Demand Side Spillovers under Different Degrees of Cooperation	FS IV 99 - 2
Horst Albach	Des paysages florissants? Une contribution à la recherche sur la transformation	FS IV 99 - 3
Jeremy Lever	The Development of British Competition Law: A Complete Overhaul and Harmonization	FS IV 99 - 4
Damien J. Neven Lars-Hendrik Röller Zhentang Zhang	Union Power and Product Market Competition: Evidence from the Airline Industry	FS IV 99 - 5
Justus Haucap Uwe Pauly Christian Wey	The Incentives of Employers' Associations to Raise Rivals' Costs in the Presence of Collective Bargaining	FS IV 99 - 6
Jianbo Zhang Zhentang Zhang	Asymptotic Efficiency in Stackelberg Markets with Incomplete Information	FS IV 99 - 7
Justus Haucap Christian Wey	Standortwahl als Franchisingproblem	FS IV 99 - 8
Yasar Barut Dan Kovenock Charles Noussair	A Comparison of Multiple-Unit All-Pay and Winner-Pay Auctions Under Incomplete Information	FS IV 99 - 9
Jim Y. Jin	Collusion with Private and Aggregate Information	FS IV 99 - 10
Jos Jansen	Strategic Information Revelation and Revenue Sharing in an R&D Race with Learning Labs	FS IV 99 - 11
Johan Lagerlöf	Incomplete Information in the Samaritan's Dilemma: The Dilemma (Almost) Vanishes	FS IV 99 - 12
Catherine Matraves	Market Integration and Market Structure in the European Soft Drinks Industry: Always Coca-Cola?	FS IV 99 - 13
Pinelopi Koujianou Goldberg Frank Verboven	The Evolution of Price Discrimination in the European Car Market	FS IV 99 - 14
Olivier Cadot Lars-Hendrik Röller Andreas Stephan	A Political Economy Model of Infrastructure Allocation: An Empirical Assessment	FS IV 99 - 15
Holger Derlien Tobias Faupel Christian Nieters	Industriestandort mit Vorbildfunktion? Das ostdeutsche Chemiedreieck	FS IV 99 - 16
Christine Zulehner	Testing Dynamic Oligopolistic Interaction: Evidence from the Semiconductor Industry	FS IV 99 - 17

Johan Lagerlöf	Costly Information Acquisition and Delegation to a "Liberal" Central Banker	FS IV 99 - 18
Ralph Siebert	New Product Introduction by Incumbent Firms	FS IV 99 - 19
Ralph Siebert	Credible Vertical Preemption	FS IV 99 - 20
Ralph Siebert	Multiproduct Competition, Learning by Doing and Price-Cost Margins over the Product Life Cycle: Evidence from the DRAM Industry	FS IV 99 - 21
Michael Tröge	Asymmetric Information Acquisition in Credit Auction	FS IV 99 - 22
Michael Tröge	The Structure of the Banking Sector, Credit Screening and Firm Risk	FS IV 99 - 23
Michael Tröge	Monitored Finance, Usury and Credit Rationing	FS IV 99 - 24
Silke Neubauer	Multimarket Contact, Collusion and the International Structure of Firms	FS IV 99 - 25
Horst Albach Thomas Brandt Holger Jakob M. A. Paradowska-Thimm Jianping Yang	Dokumentation der "Bonner Stichprobe" – Zur Datenbank der Jahresabschlüsse deutscher Aktiengesellschaften, 1960-1997	FS IV 99 - 26
Tomaso Duso	Endogenous Switching Costs and the Incentive for High Quality Entry	FS IV 99 - 29
Jos Jansen	Regulating Complementary Input Supply: Production Cost Correlation and Limited Liability	FS IV 99 - 30
Robert Greb	Internationalisierung der FuE-Tätigkeit von Unter- nehmen der Chemischen Industrie in Deutschland	FS IV 99 - 34
Suchan Chae Paul Heidhues	The Effects of Downstream Distributor Chains on Upstream Producer Entry: A Bargaining Perspective	FS IV 99 - 35
Tobias Miarka	The Recent Economic Role of Bank-Firm- Relationships in Japan	FS IV 99 - 36
William Novshek Lynda Thoman	Demand for Customized Products, Production Flexibility, and Price Competition	FS IV 99 - 37

DISCUSSION PAPERS 2000

Justus Haucap Uwe Pauly Christian Wey	Collective Wage Setting When Wages Are Generally Binding: An Antitrust Perspective	FS IV 00 - 01
Stephanie Aubert Andreas Stephan	Regionale Infrastrukturpolitik und ihre Auswirkung auf die Produktivität: Ein Vergleich von Deutschland und Frankreich	FS IV 00 - 02
Achim Kemmerling Andreas Stephan	Political Economy of Infrastructure Investment Allocation: Evidence from a Panel of Large German Cities	FS IV 00 - 03
Andreas Blume Asher Tishler	Security Needs and the Performance of the Defense Industry	FS IV 00 - 04
Tomaso Duso	Who Decides to Regulate? Lobbying Activity in the U.S. Cellular Industry	FS IV 00 - 05
Paul Heidhues Johan Lagerlöf	Hiding Information in Electoral Competition	FS IV 00 - 06
Andreas Moerke Ulrike Görtzen Rita Zobel	Grundlegende methodische Überlegungen zur mikroökonomischen Forschung mit japanischen Unternehmensdaten	FS IV 00 - 07
Rabah Amir	Market Structure, Scale Economies, and Industry Performance	FS IV 00 - 08
Lars-Hendrik Röller Johan Stennek Frank Verboven	Efficiency Gains from Mergers	FS IV 00 - 09
Horst Albach Ulrike Görtzen Tobias Miarka Andreas Moerke Thomas Westphal Rita Zobel	Documentation of the Kaisha-Database – The Annual Accounts Database of Japanese Stock Companies 1970 – 1999 With a detailed Glossary of Japanese Accounting Terminology	FS IV 00 - 10
Paul Heidhues	Employers' Associations, Industry-wide Unions, and Competition	FS IV 00 - 11
Roman Inderst Christian Wey	Market Structure, Bargaining, and Technology Choice	FS IV 00 - 12
Michael R. Baye Dan Kovenock Casper G. de Vries	Comparative Analysis of Litigation Systems: An Auction-Theoretic Approach	FS IV 00 - 13
Damien J. Neven Lars-Hendrik Röller	The Scope of Conflict in International Merger Control	FS IV 00 - 14
Damien J. Neven Lars-Hendrik Röller	Consumer Surplus vs. Welfare Standard in a Political Economy Model of Merger Control	FS IV 00 - 15
Jos Jansen	Coexistence of Strategic Vertical Separation and Integration	FS IV 00 - 16

Johan Lagerlöf	Policy-Motivated Candidates, Noisy Platforms, and Non-Robustness	FS IV 00 - 17
Pierre Mohnen Lars-Hendrik Röller	Complementarities in Innovation Policy	FS IV 00 - 18

Absender/From:				
Versandstelle - WZB Reichpietschufer 50 D-10785 Berlin				
BESTELLSCHEIN / ORD	ERFORM	Bitte schicken Sie bei Ihren Bestellungen von WZB-Papers unbedingt eine 1-DM-Briefmarke pro paper und einen an Sie adressierten Aufkleber mit. Danke.		
Bitte schicken Sie mir aus der Liste der Institutsveröffentlichungen folgende Papiere zu:		For each paper you order please send a "Coupon-Réponse International" (international money order) plus a self-addressed adhesive label. Thank You.		
Please send me the follow	ving papers from your Publica	ation List:		
Paper Nr./No.	Autor/Author + Kurz	ztitel/Short Title		